

Temi di Discussione

(Working Papers)

Exposure to media and corruption perceptions

by Lucia Rizzica and Marco Tonello







Temi di discussione

(Working papers)

Exposure to media and corruption perceptions

by Lucia Rizzica and Marco Tonello

Number 1043 - November 2015

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Pietro Tommasino, Piergiorgio Alessandri, Valentina Aprigliano, Nicola Branzoli, Ines Buono, Lorenzo Burlon, Francesco Caprioli, Marco Casiraghi, Giuseppe Ilardi, Francesco Manaresi, Elisabetta Olivieri, Lucia Paola Maria Rizzica, Laura Sigalotti, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

EXPOSURE TO MEDIA AND CORRUPTION PERCEPTIONS

by Lucia Rizzica* and Marco Tonello*

Abstract

We analyse the impact of exposure to corruption news on individuals' perceptions about the extent of the phenomenon. To this purpose, we take information on individuals' perceptions of the likelihood that corruption events may occur in everyday life and combine it with a dataset containing the number of news items related to corruption that appeared on the homepages of the websites of the 30 most widely read national and local newspapers on the day on which the individual was interviewed. Results show that increasing potential exposure to corruption news by one standard deviation causes an increase in corruption perception of about 3.5 per cent and a decrease in trust in justice effectiveness of about 5.2 per cent. We suggest that these effects are mainly driven by a persuasive mechanism rather than by a learning process so that individuals' perceptions about corruption appear to be biased by media content.

JEL Classification: D84, K42, K49.

Keywords: corruption perceptions, media, newspapers.

Contents

1. Introduction	5
2. Related literature	7
3. Data and descriptive evidence	9
3.1 Data	
3.2 Descriptive statistics	
4. Identification	
5. Results	14
6. Sensitivity and robustness checks	15
7. Potential mechanisms	19
8. Concluding remarks	
References	
Figures and tables	
Appendix: Data	
A.1 The 2014 IntermediateWave of the Italian Survey of Household Income	
and Wealth (SHIW)	
A.2 The Corruption News Database	
A.3 Data on convictions	

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research, Structural Economic Analysis Directorate, Law and Economics Division.

1 Introduction¹

While understanding the effects of corruption on the (mis)allocation of resources is at the heart of most economic and political debate, answering the question of how we can measure the extent of corruption presents a major preliminary challenge. Indeed, corruption is, by definition, a secretive act, even more so than most crimes as it does not have a clearly identified victim who may have an interest in reporting the crime. So how can we study something that we cannot measure? How well do existing measures of corruption, mostly based on individuals' perceptions, reflect actual levels of public sector corruption? Cross-national corruption perception measures have come under much theoretical and empirical scrutiny in recent years, which has serious implications for the validity and reliability of the data. Several scholars have argued that perceptions of corruption do not reflect actual levels of individuals and local conditions (Charron, 2015). Moreover, a number of recent empirical studies, mainly focusing on developing areas (Olken and Pande, 2012), have put forth evidence that outside experts' assessments of corruption correspond little, if at all, to the experience and views of citizens, thus casting a shadow on the validity and reliability of existing measures of corruption based on perceptions.

In this paper we focus on one potential channel that might bias individuals' perceptions about corruption, i.e. exposure to corruption related media content. A growing body of literature has shown that media content has the potential to affect the behavior of individuals and therefore has significant economic consequences. This is true for consumption and savings behaviors (Bertrand et al., 2010; De Paola and Scoppa, 2014), as well as for voting choices (Della Vigna et al., 2014; Barone et al., 2015), violent behaviors (Dahl and Della Vigna, 2009) and family formation decisions (Chong and La Ferrara, 2009; Bassi and Rasul, 2015). In principle, exposure to corruption news reported by the media could serve as an important tool for individuals to gain more *information* about the magnitude of the phenomenon. However, if the media reports excessively on corruption scandals and news even in the absence of real corruption events, this might lead to a *bias* in the formation of individuals' perceptions about the extent of corruption in a society. On top of that, increasing the bias in individual perceptions might result in the formation of biased beliefs through a multiplier effect.

We draw from two original data sources to uncover the causal effect of exposure to related

¹ The views expressed in this paper are those of the authors and do not involve the responsibility of the Bank of Italy. We would like to thank all the seminar participants at the 3^{rd} Giorgio Rota Conference (Einaudi Institute, Turin) and at Gotheborg University. Special thanks go to Magda Bianco, Francesco Drago, Tommaso Frattini, Nicola Gennaioli, Silvia Giacomelli, Randi Hjalmarsson, Katarina Nordblom, Ola Olsson, Giuliana Palumbo, Paolo Sestito, Friedrich Schneider, Marco Manacorda and Anna Bindler for their useful comments and suggestions. We are grateful to Fabio Bartolomeo (Directorate for Statistics, Ministry of Justice) for his invaluable help in making data on convictions available and to Giuseppe Ilardi for his assistance with the SHIW data. All errors are ours.

media content on individual perceptions of corruption and of the effectiveness of activity to combat corruption. Perceptions of corruption are derived from a set of specific questions contained in the 2014 wave of the *Italian Survey of Household Income and Wealth*, conducted by the Bank of Italy on a representative sample of about 1,800 heads of households. The survey contains several questions aimed at capturing individual perceptions of how widespread corruption is and how effective police investigations and the judicial system are in combatting it. Over the same period in which the survey interviews were conducted, we collected daily information on corruption news and scandals that appeared on the front pages of 30 online newspapers. The type of news we recorded includes not only factual reporting on corruption, such as arrests or judicial sentences for bribes, but also, for example, politicians' speeches and statements about the fight against corruption or information on the release of institutional reports on the spread of corruption across countries. Combining these two sources of data and exploiting the random scheduling of the interviews, we manage to identify the causal effect of interest.

The results show that there is a positive causal relationship between exposure to corruption news and perceptions of corruption. This relationship is stronger for the question on the survey describing the most serious cases of corruption, which individuals are least likely to have experienced directly. Interestingly, with regard to perceptions about the effectiveness of measures to combat corruption, only that on the effectiveness of the judicial system is (negatively) affected by media exposure, while that on the effectiveness of investigations is not.

Our work contributes to the literature in several respects. First, we provide evidence on media persuasion as it relates to beliefs about corruption, an aspect that has never been investigated before. We show that a causal relation exists between media coverage of corruption news and perceptions about the spread of corruption and the effectiveness of the justice system. This, in turn, provides indirect evidence that corruption indicators that are based on perceptions can be misleading, but also has potential major implications in terms of voting behavior and investment decisions. Secondly, thanks to the ampleness of our data, we are able to investigate the determinants of corruption perceptions at the micro level, and test whether characteristics such as gender, education, occupation and the frequency of interaction with public officers have an effect in shaping these perceptions. Finally, we provide some tentative evidence on the mechanisms underlying the identified effect and suggest that individuals are more affected by news that report 'claims' about corruption ('bias channel') than by those that report actual 'facts' ('learning channel').

The remainder of the paper is organized as follows. Section 2 reviews the relevant literature;

Section 3 describes the data sources used and provides general descriptive evidence; Section 4 illustrates the identification strategy and discusses the possible sources of confounding factors; Section 5 provides the baseline results; Section 6 provides a battery of robustness and sensitivity checks; Section 7 discusses the potential channels and Section 8 presents our conclusions.

2 Related literature

This paper is related to two main strands of the literature. First, it contributes to the literature on corruption measurement. As corruption is a largely unobserved phenomenon, many scholars have studied how to best uncover corruptive practices and have drawn attention to the main shortcomings and advantages of existing methodologies. Second, our work contributes to the recent and growing literature that investigates the influence of media on beliefs and perceptions. This stream of literature on *persuasion* employs approaches from psychology to incorporate aspects of bounded rationality into the agents' decision making processes. In what follows, we briefly review the evidence along these two lines found in the existing literature.

Measuring corruption. Since it is impossible to fully observe corruption, scholars have moved towards using measures that are based on subjective estimates, or *perceptions*, of corruption rather than on measuring the actual amount of bribes paid or actual thefts or misuse of public resources.² Micro founded studies seem to confirm that perceptions do contain some information about the true extent of corruption. Fisman and Miguel (2007) for instance, studied the parking practices of UN diplomats of different nationalities residing in New York and found that there is a strong positive correlation between the level of corruption in the diplomat's home country – as predicted by perception indices – and the actual amount of each diplomat's accumulated unpaid parking tickets, a proxy for propensity towards illegal behaviors. Olken (2009) examined the accuracy of corruption perceptions by comparing Indonesian villagers' reported perceptions of corruption in a road-building project in their village with a more objective measure of missing expenditures for the project. The results showed that villagers' perceptions do contain some information about the real level of corruption in the project, even if the magnitude of the perceived corruption is smaller than the objective measure and the level of perceived corruption responds little to changes in actual corruption.

²Indicators of this type range from The Economist Intelligence Unit's Business International Indicators, to the World Bank Governance Indicators, the Transparency International Corruption Perceptions Index, the Global Corruption Barometer, and the European Commission Eurobarometer. Economists have largely used these data to run cross-country regressions on various aspects: Mauro (1995); Knack and Keefer (1995) estimated the impact of corruption on growth, La Porta et al. (1999) investigated the determinants of cross-country differences in government quality and corruption, Fisman and Gatti (2002) studied the relationship between fiscal decentralization and corruption, and Fredriksson and Svensson (2003) studied that between political instability and corruption.

Other studies have questioned the reliability of perception based corruption measures, suggesting that perceptions may deviate from experience in systematic ways that may eventually overturn cross-country rankings based on perception indices. Indeed, these are likely to be affected by individual or country specific characteristics even holding the experience of corruption fixed (Banerjee and Hanna, 2012; Donchev and Ujhelyi, 2014). First, there is a problem of defining corruption. While a common and broad definition is that 'corruption is the misuse of public office for private gain' (Svensson, 2005), the boundary between what is corruption and what is not is in the end defined by law. Yet, laws differ across countries and sometimes even within countries. Second, and most importantly, there may be significant differences in cultural and social norms across and within countries so that citizens of one area may find certain practices more acceptable than citizens of another area, which would, for this very reason, be judged to be more corrupt according to opinion-based indices.³

In the light of these limitations, the most recent economic literature has moved towards developing new tools to measure the extent of corruption. Some have tried to refine surveys on actual practices in order to elicit truthful answers: Svensson (2003) gathered information from a sample of Ugandan firms about the amount of bribes they were actually paying; Olken and Barron (2009) provided evidence based on direct observation; Ferraz and Finan (2011), Brollo and Troiano (2013) and Brollo et al. (2013) used data from random audits of governmental processes and public entities in Brazil. While many of these studies have proposed more accurate and reliable methods of measuring corruption, their implementability remains limited because they are generally very costly and difficult to replicate across countries. For this reason, perception-based indices remain the main source of information for policymakers and stakeholders. In this paper we look at a channel as yet unexplored that may make these measures misleading and thus shed light on a source of volatility in perceptions that tends to disproportionately amplify differences across little and very corrupt countries or areas, i.e. media content. Indeed media content simultaneously expresses common beliefs about corruption and is a source of information on which these beliefs are based, and, for this reason it tends to polarize differences in perception across countries.

³For example, according to a survey published by the World Bank and the Government Inspectorate of Vietnam (Anderson et al., 2010), in 2010 29 per cent of Vietnamese patients of public hospitals gave their doctors an envelope with money to speed up and secure their service and of these over 75 per cent did it voluntarily without being asked to by the doctor. Scholars argue that this attitude is rooted in the Confucian gift-giving tradition for which gifts stem from gratitude. Also, a recent paper by Lee and Guven (2013) shows that more masculine societies have lower probabilities of viewing bribery as being seriously wrong and that an individual's attitude towards risk taking is among the main determinants of justification of bribes.

Media persuasion. Perceptions, or *beliefs*, about the extent of corruption are likely to affect numerous individual actions: citizens' voting choices, entrepreneurs' investment decisions, and governments' funding allocations. In the absence of being able to directly observe how much corruption there is, individuals are bound to learn about these practices from the signals they receive from more informed parties, primarily the media. A recent strand of literature has indeed focused on the persuasive power of the media on individual beliefs and behaviors (Della Vigna and Gentzkow, 2010).

The most abundant literature has looked at the effects of media content on voting behaviors (Della Vigna and Kaplan, 2007; Enikolopov et al., 2011; Barone et al., 2015; Della Vigna et al., 2014). Closely related to these papers is the work by Mastrorocco and Minale (2015), who show that reducing exposure to crime-related news decreases elderly individuals' concerns about crime. The authors argue that this change in crime perceptions is likely to have important implications for voting behavior.

But the literature on the persuasive power of the media is not limited to its effects on voting. Media content has also been shown to have an effect on violent behavior (Dahl and Della Vigna, 2009), family decisions and fertility choices (Chong and La Ferrara, 2009; Chong et al., 2012; Bassi and Rasul, 2015) and attitudes towards gambling (De Paola and Scoppa, 2014).

Our paper will investigate the impact of newspaper reports on corruption on beliefs about corruption in Italy. In addition to being an as yet unexplored channel of persuasion by the media, it is likely to have a significant impact on voting behavior, on choosing between public and private services or jobs, and on investment decisions, and can thus help us understand individual choices made in many different fields.

3 Data and descriptive evidence

We combine information on corruption perceptions and media coverage of news about corruption using two original sources of data. Corruption perceptions are collected by looking at specific questions contained in the 2014 wave of the *Italian Survey of Household Income and Wealth* (SHIW), while we built a *Corruption News Dataset* (CND) to gather information on news items about corruption reported on the front pages of a representative sample of online newspapers on the same days that the SHIW interviews were conducted.

3.1 Data

Measures of corruption perceptions. The 2014 wave of the SHIW included a representative sample of about 1,800 households.⁴ The survey was conducted by professional interviewers between January and March 2014. The questionnaire contained a theme-based section with a set of questions aimed at describing the respondents' perceptions of corruption. These questions were answered only by heads of households and thus refer to their own perceptions and not those of the other members of their households.

Specifically, respondents were presented with situations in which a hypothetical citizen is faced with a request for a bribe or similar dishonest behavior on the part of a public officer, and were asked to assign to each of these events a subjective probability of occurrence between 0 (not at all likely) and 100 (certain to happen). The questionnaires asked respondents to describe how likely they think that the following events would occur: (*i*) a public officer hints that he would accept a sum of money, a favor or a gift in exchange for providing the service; (*ii*) the citizen has to ask for the intervention of a friend or acquaintance who works in the government office in order to speed up the provision of a service; (*iii*) a corrupt public officer who has been discovered, ends up in prison. From these questions we retrieve subjective probabilities about *respondents' perceptions* with regard to three main domains, respectively: (*i*) formal corruption ('Corruption'), (*ii*) resorting to the use of private networks as a bad social norm related to corruption ('Social Norms'), (*iii*) the effectiveness of the judicial system to combatting corruption ('Justice Effectiveness').⁵

The questions focused on cases of 'petty' corruption, i.e. situations which are somehow likely to occur to citizens in everyday life and not only to businessmen and firms interacting with public officers (in contrast with 'grand' corruption, which occurs, for instance, in public tenders).⁶

Table 1 provides an overview of the corruption perceptions obtained from the survey. The descriptive pattern that emerges reassures us that the survey questions are able to capture individual expectations about corruption events. Indeed, we observe that respondents assign higher probability to corruption events that constitute less serious offenses (i.e. on average we observe that corruption is expected to occur in 36.5 per cent of the cases, while the use of connections is expected almost 50 per cent of the times). In analysing the heterogeneity of the responses based on observable characteristics we find that lower educated individuals tend to

⁵Henceforth, we will simply use the term 'corruption perceptions' to refer to the three variables. The complete translation of the SHIW Survey Questionnaire is included in Appendix A.

⁴See Appendix A for details.

⁶These definitions are borrowed from Transparency International, http://www.transparency.org/what-is-corruption/.

attach a higher probability to the occurrence of corruption events; similarly, individuals who are either not employed (inactive or unemployed) or self-employed tend to report higher perceptions of corruption than persons who are employed. Direct contact with public officers increases the perceived likelihood of corruption since those individuals who report visiting government offices more frequently also have a higher perception of corruption.

To assess how good the survey questions are overall at eliciting corruption perceptions, we compare their answers with those of similar questions posed in the Eurobarometer Corruption Report for the year 2014 (European Commission, 2014). For instance, the Eurobarometer reports that 42 percent of Italian respondents agree that 'corruption can potentially affect his/her daily life', this figure is close to the mean of the *Corruption* perception variable obtained from the SHIW (36.5 per cent).⁷ With respect to the effectiveness of law enforcement, according to the Eurobarometer report, only 27 per cent of Italian respondents (in line with the EU average) agree that 'there are enough successful prosecutions to deter people from corrupt practices'. Again, this figure is not far from the SHIW figure measuring the perception of Justice Effectiveness (15.8 per cent).

Measures of media coverage of corruption. We collected daily information on all news items related to corruption that appeared on the front page of 30 on-line newspapers (j = 1...30) for the weeks in which the SHIW interviews were conducted (i.e. between January and March 2014). We henceforth, refer to our collection as the *Corruption News Database*, CND.⁸ We selected the widely most read national newspapers (N) (including three sports newspapers) and the local newspapers with the greatest circulation (L). For each newspaper front page, we recorded the number and the type of news items containing (either in the title or in the text of the article) any of a family of keywords referring to corruption.⁹

We thus construct a measure of the potential exposure to corruption news (i.e. $News_{dp}$, eq. 1), given by the sum of two components. The first component accounts for exposure to daily news at the national level: it is given by the sum of corruption news items that appeared on day d in all national newspapers and it is therefore the same for all individuals interviewed on the same day d. The second component captures the geographical variation in exposure and is given by the sum of the corruption news items that appeared on day d in all the local newspapers

⁷According to Eurobarometer, the EU average perception is far lower, about 26 per cent, with Italy ranking just behind Spain and Greece (62 per cent), and Cyprus and Romania (57 per cent).

⁸See Appendix A for the complete list of newspapers monitored and for details on the creation of the database. Our data collection methodology is similar to that of Di Tella and Franceschelli (2011). They kept track, on a daily basis, of articles about corruption that appeared on the front page of the four most widely read national newspapers in Argentina. The aim of their paper was to study the relationship between corruption news reporting and government funding of the newspapers through advertising.

⁹See Appendix A for the list of keywords searched.

circulating in province p where the respondent i resides.¹⁰ The latter component accounts for the fact that local news items usually resonate only with respondents residing in a given area, and are not normally reported in the national press.

$$News_{dp} = \sum_{j \in N} news_{dj} + \sum_{j \in L} news_{dL}|_{d_{ip}(j \in L) > 0}$$
(1)

3.2 Descriptive statistics

Our final sample consists of 1,805 (heads of) households surveyed over 64 days between 11 January and 22 March, 2014. On average, we recorded about 12 corruption news items per day, with a peak of 39 corruption news items on one day, and a low of just one corruption news item (Table 2). The average interviewee is about 60 years-old, with half of the sample being female. Only 12 per cent of the interviewees hold a college degree and almost 56 per cent declare that they visit a government office five times per year at most, while only 7 per cent declare that they never visit a government office.

Figure 1 shows the daily variation of exposure to corruption news ($News_{dp}$, eq. 1). The vertical lines indicate the major corruption events that occurred during the survey sample period. Figure 2, displays the geographical variability across provinces¹¹ and shows that there were provinces that were incidentally more exposed than others to media coverage about corruption news during the survey period.

Figure 3 overlays the timeline of the corruption news measure and the timeline of the *Corruption* perception measure, showing our main source of identification. At first glance, we see that the two variables seem to move very closely in tandem; the vertical lines denote the most relevant corruption news published during the period of the interviews and correspond with peaks in perceptions of corruption (similar results hold for the other outcome variables).

4 Identification

The aim of our empirical analysis is to identify the effect that exposure to corruption news has on respondents' corruption perceptions. To this purpose, we focus on the following baseline specification:

$$Y_{iped} = \alpha_0 + \beta News_{dp} + \alpha_1 X_i + \alpha_2 Int_i + \varphi_e + \varphi_t + \varphi_p + \epsilon_{iped}$$
(2)

¹⁰The circulation of local newspaper j in province p ($d_{ip}(j \in L)$) must be greater than zero. Local newspaper circulation is taken from the official data of Accertamenti Diffusione Stampa srl for the year 2013. See Appendix A for details.

¹¹In Italy there are 110 provinces corresponding to the NUTS 3 classification level.

where Y_{iped} expresses the perceptions about Corruption, Social Norms, or Justice Effectiveness reported by individual *i*, residing in province *p* and being interviewed by enumerator *e* on day *d*. News_{dp} is the measure of exposure to corruption news, as obtained from eq. 1, and varies at the day and province level. X_i is a set of individual socio-demographic characteristics, such as educational level, occupational status and frequency of contact with public officers; Int_i is a set of interview-specific characteristics, including its duration and whether it was carried out in the morning;¹² φ_t represents day of the week fixed effects, φ_p province fixed effects and φ_e enumerator fixed effects.

Our research design exploits the quasi-experimental variation in the level of individual potential exposure to corruption news caused by the the random scheduling of the SHIW interviews. Interview dates are random in that there are no individual characteristics that determine at what point during the survey period a person is interviewed (Doerrenberg and Siegloch, 2014; Bassi and Rasul, 2015). Thus, we assume that random scheduling results in zero correlation between (observable and unobservable) individual characteristics and the number of corruption news items reported by the media on the day of the interview. Under this assumption, we exploit, as a main source of identification, the variation in the respondents' potential exposure to corruption news by date of interview and by province of residence.

In order to ascribe a causal interpretation to β , we must also rule out the possibility that variables omitted from our baseline specification have an effect in shaping respondents' corruption perceptions. Therefore, the inclusion of time and province fixed effects (φ_t and φ_p) is potentially important, since they allow us to control for any unobserved factor that is time or province specific.¹³ Moreover, the characteristics of the interview (Int_i) take into account observable differences in how the interviews were conducted. We control for the overall duration of the interview, for whether the interview started in the morning and for a linear time trend starting from the first day of the scheduled interviews. We also include fixed effects for the enumerators who conducted the phone interviews so as to control for any unobserved attitude of the compiler, which is constant across all the individuals surveyed.¹⁴

Our main identification assumption, i.e. the random scheduling of the interview dates with respect to observable and unobservable individual characteristics, can be at least partially tested. Table 3 shows the results from OLS regressions that verify whether the observable individual

 $^{^{12}}$ The complete list of the control variables can be found in Table 2

 $^{^{13}}$ E.g. day of the week and province fixed effects control for lower media attention on Mondays since much of the media coverage is devoted to sports and football and for a higher level of corruption perception due to the fact that a larger share of public employees work in the province where the regional governing body is located. In the robustness section we will also show that we can flexibly control for additional unobserved heterogeneity, both at the time and at the territory by time level (i.e. what we exploit for the identification), with more computationally demanding specifications without affecting our baseline estimates (see Section 6).

 $^{^{14}}$ Additional specification tests are performed in the robustness section 6.

characteristics X_i are uncorrelated with the number of news items that appeared on newspaper front pages on the day of the interview, $News_{dp}$. These regressions, which include fixed effects $(\varphi_t, \varphi_p, \varphi_e)$ and the set of control variables Int_i , do not reveal any statistically significant correlation.

A key piece of information that we do not observe is what newspapers the respondents actually read. Indeed, as shown in Appendix Figure A.1, different newspapers systematically over or under report news of corruption. Yet, as long as there is still variability over time within newspapers on the number of corruption related news items reported, and assuming that individual preferences for newspapers are sufficiently stable over time, randomly scheduling the interviews also makes it possible to overcome this concern.

Finally, because we do not observe the exact level of each respondent's exposure to media since we do not know whether respondents actually read any on-line newspapers on the day of the interview, we will refer to a *potential* exposure and we will interpret parameter β as an *Intention To Treat* effect (ITT).

5 Results

We run OLS regressions following our baseline specification (eq. 2) with robust standard errors clustered at the level of the day of the interview (Della Vigna and Kaplan, 2007) and report our baseline results in Table 4. Columns (1), (4) and (7) show the raw correlation between the news measure and, respectively, the *Corruption, Social Norms* and *Justice Effectiveness* perceptions, while in columns (2), (5) and (8) we add the individual characteristics (X_i) , and in columns (3), (6) and (9) we include the province and the day of the week (*dow*) fixed effects (φ_t and φ_p) and the interview controls (*Int_i* and φ_e).

Focusing our comments on the full specification in columns (3), (6) and (9), we observe that media coverage of corruption news has a positive and statistically significant effect on *Corruption* perceptions, and a negative and statistically significant effect on *Justice Effectiveness* perceptions. The size of the effect is also non-negligible: increasing media coverage by one additional corruption news item determines an increase in the perceived likelihood of a citizen being asked a bribe by a public officer (i.e. *Corruption* variable, column (3)) by 0.17 percentage points (about 0.5 per cent) and decreases the perceived likelihood that a corrupt public officer would be sentenced to prison (i.e. variable *Justice Effectiveness* variable, column (9)) by 0.11 percentage points (about 0.7 per cent). To generalize, increasing corruption news items by one standard deviation (i.e. 7.49 news) causes a 1.27 percentage points (3.5 per cent) increase in *Corruption* perceptions and a 0.82 percentage points (5.2 per cent) decrease in *Justice* *Effectiveness* perceptions.

Perceptions of social norms, as opposed to corruption or justice effectiveness, are not influenced by media coverage of corruption news items: the results for *Social Norms* show the same positive sign as *Corruption* perceptions, but are never statistically significant. This could be either because they are not perceived to be related to corruption related news, or because social norms are widely accepted and well established in the individual's mind and thus not susceptible to influence by the media in the short run.

With respect to individual characteristics, in focusing on column (3) we observe that corruption perceptions are negatively correlated with age, educational level (i.e. holding a high school diploma) and public sector employment, while gender, employment status and frequency of contact with public officers do not seem to have any significant direct influence. This piece of evidence also makes a relevant contribution to the existing literature, as very few papers have managed to provide an analysis of the determinants of the perceptions of corruption based on individual micro data, especially for developed countries (Mocan, 2008; Lee and Guven, 2013).

The use of the online versions of newspapers is an important innovation in the media and persuasion literature. While it is true that media might also report on corruption through other means, such as TV, radio or the print version of newspapers, it is also true that the type of news reported should be correlated. Therefore we do not expect our estimates to be biased because of the limited set of information sources that we consider, but rather our estimated effects will implicitly include the effects of TV or print journalism coverage of corruption related news (Eisensee and Strömberg, 2007).

6 Sensitivity and robustness checks

In this section we perform a vast array of sensitivity and robustness checks to corroborate our results and to show that the specification chosen is appropriate for handling potential threats to identification.

Measures of exposure to corruption related news. In columns (1) to (4) of Table 5 we show the results of a set of specification tests that make use of several variants of our main variable of interest (i.e. $News_{dp}$). In Figure A.1 in the Appendix, we see that the total number of corruption news items appearing in two newspapers (i.e. Il Fatto Quotidiano and La Stampa) is considerably larger than the average. In the specification of column (1) of Table 5, we therefore treat these two newspapers as outliers and exclude them from the computation of the $News_{dp}$ variable. The results do not change significantly from the baseline (i.e. Table 4, columns (3), (6) and (9)). We then compute a three day moving average of $News_{dp}$ (column (2))

and a cumulative version of $News_{dp}$ (column (3)), which is the sum of corruption news items that appeared on the day of and in the two days prior to the interview for each respondent.¹⁵ We use these two measures to take into account that the respondents' exposure might also be influenced by all the news heard in the days leading up to the interview. The results do not change significantly from the baseline, although the magnitude of the coefficients shrinks when considering the cumulative news measure. This seems to point to the fact that only the news reported on the same day as the interview matters in influencing perceptions. We will come back to this point of the persistence of the effects further on.

The $News_{dp}$ variable does not attach a different weight to the news based on the observable characteristics of the newspaper. However, it could be reasonable to assume that news items that appeared in the most widely read newspapers produce a greater echo than newspapers with smaller circulations, and thus have greater power to influence the respondents' perceptions. For this reason we compute a weighted version of the $News_{dp}$ variable, which simply sums each news item as weighted by the newspaper's relative circulation (d_j) .¹⁶ The results in column (4) imply that a one standard deviation increase in the weighted news measure causes a 0.94 percentage points increase in *Corruption* perceptions and an approximately 0.94 percentage points decrease in *Justice Effectiveness* perceptions.

Timing of the interview. We then perform additional sensitivity checks to rule out the possibility that respondents interviewed over the weekend (Saturday and Sunday) may be systematically different from the rest of the sample and be exposed to a different type of media coverage than during week days (for instance, because of sporting events, which usually take up a lot of newspaper front page space on the weekends). The inclusion in the main specification of the day-of-the week fixed effects is intended to account for these systematic differences. In Table 5 we go a step further and estimate the baseline model by excluding all respondents who were interviewed on week-end days (about 8 per cent of the total sample, column (5)) and by linking the news exposure of the previous day to all respondents who completed the interview before 11 PM ($News_{(d-1)p}$, column (6)).¹⁷ The results obtained are still in line with the baseline

¹⁵Formally, the three days moving average of $News_{dp}$ is given by $News_{dp}^{MA} = \sum_{d=-1}^{1} News_{dp}$, while the cumulative version is equal to $News_{dp}^{CUM} = \sum_{d=0}^{-2} News_{dp}$, where d = 0 is the day of the interview. ¹⁶A newspaper's total circulation is measured by the copies sold, the free copies distributed and subscriptions

¹⁶A newspaper's total circulation is measured by the copies sold, the free copies distributed and subscriptions (data published for the year 2013 by Accertamenti Diffusione Stampa srl). The relative circulation of each newspaper j is given by the ratio between its own circulation and the total circulation of the 30 newspapers in the sample. Notice that in this case we simply sum the weighted news measure across all national and local newspapers. Attempts to use alternative indicators to compute the weights (such as the number of paper copies sold and the number of digital copies sold) or using different weights for the local news component do not generate different results.

¹⁷This latter test is also important because, according to the survey protocol of the CND (see Appendix A), we performed our Boolean search on the online newspapers every day starting from 2 PM. Note, however, that the variable *Morning* in the baseline specification implicitly takes this into account.

specification.

Unobserved heterogeneity. A major threat to our identification strategy is that unobserved factors at the province by day level (other than the exposure to corruption news) might affect respondents' perceptions. The limited number of respondents in the SHIW sample makes it computationally impossible to include province-by-day fixed effects in our specification as these cells do not contain enough observations to perform the estimates. To flexibly control for unobserved heterogeneity at the territory-by-time level, we include province-by-week fixed effects: the results are shown in column (8) of Table 5 (while column (7) shows the estimates with inclusion of the week fixed effects only). Given the plausible assumption that any large scale local event (for example, a demonstration against the Mafia in province p in day d) that could have an effect on individuals' perceptions (in this example, reasonably those related to *Justice Effectiveness*) is also likely to have some short run effect that could plausibly extend beyond the precise day of the week on which the interview occurs, performing this robustness check should reassure us that our results are not driven by relevant unobserved factors at this level.

Endogeneity of News_{dp}. Exposure to certain types of news may be endogenous if editor and reader preferences are correlated so that readers choose which newspaper to read depending on its coverage of a certain topic, such as corruption (Eisensee and Strömberg, 2007). Yet, in our case, even if there was this assortative mating between readers and newspapers, the number of corruption-related news items to which a reader is exposed would remain exogenous. This happens because, in order to identify the causal effect, we use the daily variation in the number of news items reported, so that as long as editor and reader preferences remain sufficiently constant over time, the number of news to which a reader is exposed each day is exogenous. Over the survey period none of the newspapers in the sample changed editor or owner and the data on newspaper sales and on website visits show very little daily variation. Nevertheless, we run a test to rule out that some sorting of readers across newspapers is driving our results. We thus follow Olken (2009) and Leon et al. (2013) and run our main regressions including a variable that serves as a proxy for an individual's preferences and, in particular, for his propensity to care about corruption issues. The variable we employ is a question from the SHIW that asks respondents how serious they think the problem of tax evasion is. The results in column (9) of Table 5 show that the estimated coefficients for the β remain unaffected.

Persistence of the effects. We test whether media exposure has a persistent effect on corruption perceptions. In particular, we estimate a variation of equation 2, in which we focus on the effects of corruption news that appeared in the newspapers t days after the interview,

controlling for the number of news items that appeared on the day (d) of the interview:

$$Y_{idpe} = \alpha_0 + \beta News_{dp} + \beta_{lag} News_{(d-t)p} + \alpha_1 X_i + \alpha_2 Int_i + \varphi_t + \varphi_p + \varphi_e + \epsilon_{idpe}$$
(3)

This specification makes it possible to obtain estimates for the contemporaneous potential exposure to corruption news (β) and for the effects of the lagged potential exposure (β_{lag}), allowing us to analyze whether a memory process exists (i.e. there is persistence in the effects) and whether lagged exposure has a significant impact on current perceptions (Dahl and Della Vigna, 2009). We plot the estimated coefficients ($\hat{\beta}$, i.e. the contemporaneous exposure and $\hat{\beta}_{lag}$, i.e. the lagged exposure) and the corresponding 90 per cent confidence intervals in Figure 4, considering a time span of up to 30 days for the lag $News_{(d-t)p}$. The estimate of the contemporaneous exposure on day = 0 corresponds to the baseline estimate of Table 4 column (3).

Both the contemporaneous and the lagged exposure show large confidence intervals and are not statistically significant up to day eight. This is plausibly due to a high autocorrelation component that does not allow us to disentangle the two separate effects in the very short run. If we consider longer time lags (between 10 and 30 days), the estimates for the lagged variable are never statistically significant, while those for contemporaneous exposure are generally positive and significant at the 90 per cent confidence level. The timing of the effects shows that there is no persistence in the effects of news exposure on corruption perceptions, since only the contemporaneous exposure has a significant impact. Again, the results for the *Justice Effectiveness* and *Social Norms* variables, albeit confirming the general findings, are less precisely estimated.¹⁸

Placebo regressions. As a final robustness check, we perform a placebo exercise. Specifically, we randomly assign the respondents to the dates of the interviews, and re-estimate our baseline OLS specification. We repeat the random draw and the subsequent estimation for up to 1,000 replications, as in a Monte Carlo simulation process, and then average the estimated coefficients and standard errors over the number of replications. The results are depicted in Figure 5. We do not find any statistically significant effect, and as the number of replications increases, the estimated average coefficients converge to zero, providing reassurance that the effects we estimate in our baseline analysis are convincingly capturing the causal effect of media exposure on corruption perceptions.

¹⁸These results are not reported but are available from the authors upon request.

7 Potential mechanisms

In this section, we try to shed light on the mechanisms underlying the effects detected above. We believe that there are two important potential channels that may explain the influence of news coverage on corruption perceptions: on the one hand it may be that readers acquire new information from the media about a phenomenon that they do not observe directly, in which case the news on corruption would trigger a *learning* mechanism; on the other hand, to the extent that the news items reported by the media do not contain additional informational in that they do not reveal any new corruption facts to their readers, the effect of these news items on respondents' preferences should instead be interpreted as *bias*.

Facts vs. claims. In the attempt to disentangle these two mechanisms and understand which of the two is driving our results, we split our news into two categories: on the one hand 'facts', i.e. corruption news reporting on convictions, arrests or investigations; and on the other hand 'claims', i.e. corruption news items that are not linked to any specific fact. Indeed, the news items related to corruption that we collected were not always mere accounts of some corruption investigation, arrest or conviction, but were sometimes reports of some statement by politicians, international organizations or other public figure about the extent of corruption and the importance of fighting it. Clearly both types of news are likely to affect individual perceptions of the phenomenon, but we will argue that the first type of news, the facts, is a signal to readers, allowing them to revise their beliefs through a learning process. Claims, instead, tend to persuade readers without adding any new piece of information about the phenomenon, thus biasing beliefs. In other words, the two types of news represent two types of signals: good signals, which convey information, and bad signals, which instead decrease the accuracy of individuals' beliefs.

Figures 6 and 7 show the front pages of two major national newspapers reporting respectively a corruption 'fact' and a 'claim'. In the first case, the news refers to the start of a trial for alleged corruption against former Prime Minister Silvio Berlusconi, who was accused of having paid money to witnesses in another trial against him. This occurrence of corruption was unknown to readers before the news was reported in the newspaper and thus we classify this as a fact that reveals new information about the actual extent of corruption. The second news item, instead, is about the issue of a report by the European Commission about the extent of corruption across member states. The figures given in the report were based on past facts, and thus did not add any new information on the extent of corrupt practices. Moreover, the estimates of the European Commission were revealed to be erroneous a few days later (Polo, 2014), so that we can confidently classify this as a bad signal. As a first attempt to understand whether the estimated effects prevalently originate from a learning process or just reflect biased reporting, we thus run our baseline regressions excluding the days in which the main facts and claims occur. In Figure 1 we show that two main corruption claims occur in the survey period: the issuing of the EU Report on Corruption in the EU Member States (letter c in the figure) and the declarations against corruption upon settlement of the new Prime Minister Matteo Renzi (letter e in the figure). During the survey period there were also two days when the news on the EU Report overlapped with other news about a corruption scandal involving bribes in the Italian National Space Agency (letter d in the figure). These two days thus represent a combination of the two types of mechanisms and will be excluded from the analysis. Columns (1), (4) and (7) in Table 6 report the estimates of our baseline specification excluding these two days (with a peak of facts and claims), while in columns (2), (5) and (8) we exclude the days in which only claim news appeared on the newspapers (for a total of five days), and in columns (3), (6) and (9) we exclude a corresponding number of days in which we recorded the main occurrences of fact news.

Excluding the respondents interviewed on the five days when only claims appeared in the newspapers, we do not find any statistically significant effect of corruption news on *Corruption* perceptions (column (2)). Conversely, if we exclude the five days when the main facts items were reported (column (3)), we still find a positive and statistically significant effect, which almost doubles in magnitude as compared to the baseline in column (1). We interpret this piece of evidence as suggesting that, in our sample, *Corruption* perceptions are mainly affected by claims. The results in Table 6 are instead inconclusive for the *Justice Effectiveness* and *Social Norms* perceptions, as all the specifications show a negative, but not statistically significant, correlation.

Using convictions as a proxy for facts. As a second step in digging into the mechanisms that underlie the estimated effects, we used data from the Directorate for Statistics of the Ministry of Justice that report the number of individuals convicted for corruption-related crimes in every court district and on every day during the survey period.¹⁹ The data on convictions can be used as a proxy for the occurrence of corruption facts in a certain area and on a given day.

Figure 8 compares the daily variation in the total number of individuals convicted with the most reverberating news related to corruption (either facts or claims). Two main aspects are worth noting: first, claim news (i.e. the vertical lines denoted by letters c and e) never coincides with peaks in the convictions measure (this is also true for the mixed event indicated by letter d);

¹⁹For details on data on convictions see Appendix A.

conversely, in most of the cases (letters b, f, g) fact news coincides with peaks in the convictions measure. This is not the case where the corruption events covered by the media are related to arrests or investigations (as compared with convictions, letter a).²⁰

In Table 7 we first analyze whether the number of individuals convicted for corruption-related crimes on each day in each court district ($Convictions_{dc}$) has any effect on respondents' perceptions. As for the variable $News_{dp}$ in the main analysis, the conviction data on are linked to each respondent based on the day of the interview d and on the court district c in which the respondent resides.²¹ The results in column (1) show that corruption convictions have no effect on any dimension of the respondents' perceptions. Given that news on convicted individuals might take some time to appear on the newspapers' home-pages, in columns (2) to (4) we perform several robustness checks using alternative specifications for the $Convictions_{dc}$ variable, which, however, do not alter the results.

Finally, in columns (5) and (6) we augment the specification that include the $News_{dp}$ variable. Under the assumption that the number of individuals convicted of corruption-related crimes is a good proxy for the real level of corruption, the effect of potential exposure to corruption news is also given net of the actual level of corruption reported for each area and day for this specification. It turns out that the effect of potential exposure to corruption news is not influenced by the number of convictions: estimates for perceptions on *Justice Effectiveness* are even more precise once we account for the time variant part of the actual level of corruption in the area.

In turn, the results presented in Tables 6 and 7 suggest that the effects of exposure to corruption news are mainly driven by news items reporting claims rather than facts. Indeed, we find that perceptions are not influenced by corruption facts, as proxied by the number of convictions that occurred on the days near the interview.

Heterogeneous effects. As a final piece of evidence concerning the underlying mechanisms, we investigate how the effects of exposure to corruption news vary depending on certain individual characteristics that may serve as proxy for the individual's level of knowledge of the phenomenon. In Table 8 we split the sample by civil servant status (whether or not the individual works in a government office), self-employment status (whether or not the individual is self-employed), and frequency of visitation of government offices (whether or not the individual is a frequent visitor) and run separate regressions on each subsample.

 $^{^{20}}$ Unfortunately, because of privacy restrictions on the convictions database, we are not able to link convictions to the corresponding news items (if any). Figure 8 shows that some of the convictions were also reported in the news, but only if involving some public figure or politician. In general, the media tends to under report convictions.

²¹Robustness checks using alternative measures, such as the total number of persons convicted on each day, do not change the results (available from the authors upon request).

We find that respondents who are potentially *less informed* about corruption and/or *less* exposed to its risk (i.e. private sector workers, employees, people who do not frequently visit government offices) are those who react most strongly to corruption news exposure. Conversely, respondents who are potentially more informed and/or more exposed (i.e. frequent visitors of government offices, civil servants, self-employed) do not appear to be influenced by the content of the news. Again the estimates are less precise for Justice Effectiveness and there is no statistically significant effect on the ones for Social Norms.

Finally, we investigate the impact of previously accumulated knowledge of the phenomenon and news by performing the same heterogeneity analysis, excluding the days when the major facts or major claims occurred, along the lines of the analysis performed for Table 6. Focusing on *Corruption* perceptions, the results reported in Table 9 are more precisely estimated when the days with the major facts are excluded and the effects remain concentrated among less informed/exposed individuals. There are no statistically significant effects on any group when the days with the major claims are excluded.

We believe that these findings contain two main messages. First, respondents who are potentially less informed about corruption practices and less exposed to corruption in their work experience are those who react more strongly to exposure to corruption news. Moreover, they appear to be mostly affected by claims about corruption rather than facts, thus suggesting that the persuasive mechanism is stronger when individuals' beliefs are less accurate. Secondly, from a policy perspective, our results suggest that the opinions of stakeholders or informed individuals should be preferred, as compared to non-qualified citizens, to build corruption indicators based on perceptions, as these individuals are considerably less affected by media content.

8 Concluding remarks

The question of how widespread corruption is and what areas or countries are the most affected by it is an open and compelling one. Most measures used to compare and rank countries are based on surveys that collect individuals' subjective perceptions about the extent of the phenomenon. Yet these measures are likely to contain errors due to imperfect information or bias in perceptions that may alter individual responses. These errors may be more severe among certain types of respondents or in certain areas or countries where individual perceptions happen to be more malleable and volatile. This paper attempts to shed light on the sensitivity of individual perceptions of corruption to media content. For this reason, this paper stands in relation to the most recent literature on media persuasion, examining a novel means by which media can influence individual behaviors. We use two original data sources that respectively contain information about the number of news items related to corruption that appeared in the major Italian online newspapers over the period January-March 2014 and about the level of corruption perceived by around 1,800 individuals interviewed during the same period. Incorporating these corruption perception questions into an extensive, established household survey allows us to link perceptions to a large set of individual characteristics. Our strategy in identifying a causal parameter (an intention-to-treat effect) for exposure to corruption-related news on the respondents' corruption perceptions exploits the random scheduling of the interviews, controlling for the characteristics of the respondent and of the interview, as well as unobserved time and territorial heterogeneity.

We find that media content affects the perceptions of individuals, especially regarding those phenomena that they are least likely to have experienced directly. In particular, we find that increasing individual exposure to corruption news on the day of the interview by one standard deviation determines an increase in the stated likelihood of being asked for a bribe by a public official of about 3.5 per cent over the mean and a decrease in the perceived level of effectiveness of justice of about 5.2 per cent. The effects are robust to alternative specifications and robustness checks, but generally are more precisely estimated for corruption perceptions. The perceptions on social norms instead are not affected; this may be because they relate to individual beliefs that do not vary in the short run or because they are not perceived as corrupt behaviors and, therefore, as related to the news. We also try to disentangle which type of economic mechanism underlies our results. Despite being far from conclusive, our evidence suggests that individuals are more affected by claims of public figures and institutions than by the mere reporting of corruption facts. We interpret this to suggest that the prevailing mechanism is more one of *bias* than of *learning*. However, more accurate data are still needed to provide a final answer to this question.

From a policy perspective, our work has important implications since it highlights certain pitfalls in using existing measures of corruption that are based on citizens' perceptions. Since these indices have been shown to have a significant influence on the decisions of economic agents, shaping, for example, both domestic and foreign investments, our results are relevant at a macro level, suggesting that cross-country comparisons based on perception indices are to be interpreted (and built) with caution. Moreover, there should be greater emphasis on the disclosure of these measures. Indeed, we show that advertising the results of surveys based on individual perceptions produces a kind of 'snowball effect' in that this news amplifies the pre-existing differences in perceptions.

In conclusion, while it is certainly true that existing cross-national indices reflect a certain

level of corruption present in each country, it is less obvious that they are able to express the exact level of corruption in a country, and as a consequence, the exact cross-country ranking. We provide evidence that media content biases individuals' perceptions of corruption and that this effect can be heterogeneous across different types of respondents, with those who are plausibly less informed about corruption and less exposed to its risks being more likely to be influenced by media content. This factor should be taken into account when designing surveys, and further research should be carried out in order to develop better measures.

References

- Anderson, J. H., M. D. Alcaide Garrido, and T. Thi Phung (2010). Vietnam development report2010 : modern institutions. Development Policy Review (DPR) 52244, World Bank.
- Banerjee, A. and R. Hanna (2012). *Corruption*. The Handbook of Organizational Economics. Princeton University Press.
- Barone, G., F. D'Acunto, and G. Narciso (2015). Telecracy: Testing for channels of persuasion. American Economic Journal: Economic Policy forthcoming.
- Bassi, V. and I. Rasul (2015). Persuasion: A case study of papal influences on fertility preferences and behavior. Mimeo, University College London.
- Bertrand, M., D. Karlan, S. Mullainathan, E. Shafir, and J. Zinman (2010). What's Advertising Content Worth? Evidence from a Consumer Credit Marketing Field Experiment. The Quarterly Journal of Economics 125(1), 263–305.
- Brollo, F., T. Nannicini, R. Perotti, and G. Tabellini (2013). The Political Resource Curse. American Economic Review 103(5), 1759–96.
- Brollo, F. and U. Troiano (2013). What Happens When a Woman Wins an Election? Evidence from Close Races in Brazil. MPRA Paper 52244, University Library of Munich, Germany.
- Charron, N. (2015). Do corruption measures have a perception problem? assessing the relationship between experiences and perceptions of corruption among citizens and experts. *European Political Science Review*, 1–25.
- Chong, A. and E. La Ferrara (2009). Television and Divorce: Evidence from Brazilian Novelas. Journal of the European Economic Association 7(2-3), 458–468.
- Chong, A., E. La Ferrara, and S. Duryea (2012). Soap Operas and Fertility: Evidence from Brazil. American Economic Journal: Applied Economics 4(4), 1–31.
- Dahl, G. and S. Della Vigna (2009). Does movie violence increase violent crime? The Quarterly Journal of Economics 124(2), 677–734.
- De Paola, M. and V. Scoppa (2014). Media exposure and individual choices: Evidence from lottery players. *Economic Modelling* 38, 385–391.
- Della Vigna, S., R. Enikolopov, V. Mironova, M. Petrova, and E. Zhuravskaya (2014). Cross-Border Media and Nationalism: Evidence from Serbian Radio in Croatia. American Economic Journal: Applied Economics 6(3), 103–32.

- Della Vigna, S. and M. Gentzkow (2010). Persuasion: Empirical Evidence. Annual Review of Economics 2(1), 643–669.
- Della Vigna, S. and E. Kaplan (2007). The Fox News Effect: Media Bias and Voting. The Quarterly Journal of Economics 122(3), 1187–1234.
- Di Tella, R. and I. Franceschelli (2011). Government advertising and media coverage of corruption scandals. American Economic Journal: Applied Economics 3, 119–151.
- Doerrenberg, P. and S. Siegloch (2014). Is soccer good for you? the motivational impact of big sporting events on the unemployed. *Economics Letters* 123(1), 66 69.
- Donchev, D. and G. Ujhelyi (2014). What Do Corruption Indices Measure? *Economics and Politics* 26(2), 309–331.
- Eisensee, T. and D. Strömberg (2007). News droughts, news floods, and us disaster relief. *The Quarterly Journal of Economics* 122(2), 693–728.
- Enikolopov, R., M. Petrova, and E. Zhuravskaya (2011). Media and political persuasion: Evidence from russia. *American Economic Review* 101(7), 3253–3285.
- European Commission (2014). Corruption Report. Special Eurobarometer 397, European Commission.
- Ferraz, C. and F. Finan (2011). Electoral Accountability and Corruption: Evidence from the Audits of Local Governments. American Economic Review 101(4), 1274–1311.
- Fisman, R. and R. Gatti (2002). Decentralization and corruption: evidence across countries. Journal of Public Economics 83(3), 325–345.
- Fisman, R. and E. Miguel (2007). Corruption, Norms, and Legal Enforcement: Evidence from Diplomatic Parking Tickets. *Journal of Political Economy* 115(6), 1020–1048.
- Fredriksson, P. G. and J. Svensson (2003). Political instability, corruption and policy formation: the case of environmental policy. *Journal of Public Economics* 87(7-8), 1383–1405.
- Knack, S. and P. Keefer (1995). Institutions And Economic Performance: Cross-Country Tests Using Alternative Institutional Measures. *Economics and Politics* 7(3), 207–227.
- La Porta, R., F. Lopez-de Silanes, A. Shleifer, and R. Vishny (1999). The quality of government. Journal of Law, Economics, and Organization 15(1), 222–279.

- Lee, W.-S. and C. Guven (2013). Engaging in corruption: The influence of cultural values and contagion effects at the micorlevel. *Journal of Economic Psychology* 9, 287–300.
- Leon, C., J. Araa, and J. de Leon (2013). Correcting for scale perception bias in measuring corruption: an application to chile and spain. *Social Indicators Research* 114(3), 977–995.
- Mastrorocco, N. and L. Minale (2015). Information and crime perceptions: Evidence from a natural experiment. Mimeo, University College London.
- Mauro, P. (1995). Corruption and growth. The Quarterly Journal of Economics 110(3), 681–712.
- Mocan, N. (2008). What dermines corruption? international evidence from microdata. *Economic Enquiry* 46(4), 493–510.
- Olken, B. A. (2009). Corruption perceptions vs. corruption reality. Journal of Public Economics 93, 950–964.
- Olken, B. A. and P. Barron (2009). The Simple Economics of Extortion: Evidence from Trucking in Aceh. *Journal of Political Economy* 117(3), 417–452.
- Olken, B. A. and R. Pande (2012). Corruption in Developing Countries. Annual Review of Economics 4(1), 479–509.
- Polo, M. (2014). Corruzione: come mai IItalia sta peggio? Technical report, LaVoce.info.
- Svensson, J. (2003). Who Must Pay Bribes And How Much? Evidence From A Cross Section Of Firms. The Quarterly Journal of Economics 118(1), 207–230.
- Svensson, J. (2005). Eight Questions about Corruption. Journal of Economic Perspectives 19(3), 19–42.

Figures



Notes: the timeline shows the average media exposure (as obtained from eq. 1) by date. The vertical lines denote the occurrence of the major corruption events, solid lines denote the main corruption facts, dashed lines denote the main corruption *claims*; the dash-dot line denotes a date on which two main claims and facts overlap. The lines denote, in chronological order: (a) embezzlement cases involving the Piedmont governing body; (b) the so-called 'Rubi ter trial' in which Silvio Berlusconi (Italy's former prime minister) was accused of corruption; (c) the issuing of the EU Anti-Corruption Report; (d) a case of bribery in the Italian Space Agency and news on the EU Anti-Corruption Report; (e) the appointment of a new prime minister (Matteo Renzi) with statements about the fight against corruption; (f) the so-called "Maugeri trial' in which members of the Lombardy governing body were formally accused of corruption and embezzlement; (g) the case of corruption discovered in the Lombardy public health system. **Source**: the authors' calculations based on data from the Corruption News Database (CND).



Figure 2 Corruption news exposure: geographical variation

Notes: the map shows the average of the measure of media exposure by province as expressed by eq. 1. The darker the

Notes: the map shows the average of the measure of media exposure by province as expressed by eq. 1. The darker the area, the more corruption news items reported in local newspapers. **Source**: the authors' calculations based on data from the Corruption News Database (CND).



Figure 3 Corruption news exposure and corruption perceptions: the overlap of the daily timeline

Notes: the timeline shows the average media exposure (as obtained from eq. 1) and *Corruption* perceptions by date. The vertical lines denote the occurrence of events listed in Figure 1. **Source**: the authors' calculations based on data from the Italian Survey of Household Income and Wealth (2014) and the Corruption News Database (CND).

Figure 4 Timing of the effect: corruption



Notes: the graph shows the coefficients (dots) and the confidence intervals at the 90 per cent level (dashed lines) of the effects of contemporaneous (upper panel) and lagged (bottom panel) exposure to news. The results are obtained from the regression specification that includes the same control variables and fixed effects as in the baseline specification of Table 4, columns (3), (6) and (9). The contemporaneous effect on day = 0 corresponds to the baseline estimate of Table 4 column (3). **Source**: the authors' calculations based on data from the Directorate for Statistics of the Italian Ministry of Justice.



Figure 5 Robustness: placebo Monte Carlo simulations

Notes: the figures in Panel (A), (B) and (C) show, respectively, the results of the placebo Monte Carlo simulation for the *Corruption, Social Norms* and *Justice Effectiveness* variables. The dots correspond to the average coefficient estimates for the parameter β from eq. 2 computed over a given number of replications indicated on the horizontal axes; the vertical lines denote the average confidence intervals at the 95 per cent level computed over the corresponding number of replications. Source: the authors' calculations based on data from the Italian Survey of Household Income and Wealth (2014) and the Corruption News Database (CND).

Figure 6 Example of a major corruption 'fact'



Figure 7 Example of a major corruption 'claim'





Figure 8 Persons convicted for corruption-related crimes by date

Notes: the timeline shows the total number of persons convicted for corruption-related crimes by date. The vertical lines denote the occurrence of events listed in Figure 1. **Source**: the authors' calculations based on data from the Directorate for Statistics of the Ministry of Justice.

Tables

	Ν	Corruption	Social Norms	Justice Effectiveness
Occupation status				
Employee	532	33.77 (28.22)	47.47 (31.1)	16.31 (21.65)
Self-employed	136	37.07 (27.66)	50.05(29.22)	12.28 (18.77)
Unemployed or out of the labour force	1137	37.65(28.97)	49.63(31.11)	15.99(20.75)
Education level				
Higher education	841	33.9(27.47)	48.91(30.67)	15.35(20.69)
Lower education	964	38.7(29.55)	49.13(31.25)	16.2(21.07)
Contacts with Public Officers				
Lower than the median	965	34.13(27.5)	46.54(30.35)	14.83(20.13)
Higher than the median	840	39.14(29.8)	51.88(31.45)	16.91(21.69)
Total	1805	36.46 (28.69)	49.03 (30.97)	15.8 (20.89)

Table 1 Descriptive statistics: corruption perceptions from the SHIW data

Notes: the figures report the mean coefficients and the standard deviations (in parentheses) of the answers to the SHIW Survey Questions designed to elicit individuals' corruption perceptions (see Appendix A). We define individuals according to the following characteristics: *Employees* include those with either a tenure or a fixed term contract, *Unemployed or out of the labour force*; individuals are classified as having a *higher education* they hold at least a high school diploma; individuals with *Contacts with Public Officers* below the median are those who declare that they have had five or fewer contacts per year. **Source**: Italian Survey of Household Income and Wealth (2014).

	Table 2			
Descriptive statistics:	dependent	and	$\operatorname{control}$	variables

Variable name	Mean	Sd	Max	Min	Ν
Dependent variables					
Corruption	36.46	28.69	100	0	1805
Social Norms	49.03	30.97	100	0	1805
Justice Effectiveness	15.8	20.89	100	0	1805
Independent variables					
News	12.1	7.49	30	1	889
Female	0.5	0.5	1	0	1805
Age	60.25	15.38	95	18	1805
Junior School	0.53	0.5	1	0	1805
High School	0.35	0.48	1	0	1805
College	0.12	0.32	1	0	1805
Employee	0.29	0.46	1	0	1805
Self employed	0.08	0.26	1	0	1805
Unemployed or out of the labour force	0.63	0.48	1	0	1805
Civil servant	0.1	0.3	1	0	1805
POF: never	0.07	0.25	1	0	1805
POF: at most 5 times py	0.56	0.5	1	0	1805
POF: at most 10 times py	0.21	0.41	1	0	1805
POF: more than 10 times py	0.16	0.37	1	0	1805
Morning	0.33	0.47	1	0	1805
Duration	39.47	14.39	159	15	1805
Persons convicted for corruption-related crimes	0.27	0.65	4	0	611

Notes: Corruption, Social Norms and Justice Effectiveness are the dependent variables expressing individual perceptions (in terms of subjective probabilities) as defined based on the SHIW (see Appendix A for details on their formulation); News indicates corruption news exposure (per day, per province) as obtained from eq. 1; Female is a dummy variable equal to 1 if the individual is female; Age is a variable indicating the individual's age in years; Junior School, High School, College are dummy variables equal to 1 if the maximum level of education completed by the individual is, respectively, a Junior School Diploma, the High School Diploma, the College Degree or any higher educational qualification; Employee is a dummy variable equal to 1 if the individual holds a permanent or a fixed-term position as an employee; Self-employed is a dummy variable equal to 1 if the individual is self-employed; Unemployed or out of the labour force is a dummy variable equal to 1 if the individual is self-employed; Unemployed or out of the labour force is a dummy variable equal to 1 if the individual is self-employed; Unemployed or out of the labour force is a dummy variable equal to 1 if the individual is self-employed; Unemployed or out of the labour force is a dummy variable equal to 1 if the individual is either retired or in search of a job; Civil servant is a dummy variable equal to 1 if the individual is enployed in the public sector; the frequency of the individual's contacts with the public officers (POF) is indicated by dummy variables for the categories: POF: never, POF: at most 5 times per year (py), POF: at most 10 times per year (py); Norning is a dummy variable equal to 1 if the interview was started in the morning; Duration indicates the duration of the interview (in minutes); Persons convicted for corruption-related crimes is the number of persons convicted for corruption related crimes on the day of the interview (by court district) Source: Italian Survey of Household Income and Wealth (SHIW), Corruption New

	Dep.	Variable:	News
Female	-0.00	0.15	-0.03
	(0.33)	(0.29)	(0.30)
Age	0.01	-0.00	-0.02
	(0.02)	(0.01)	(0.01)
High School	0.72^{*}	0.32	0.40
	(0.41)	(0.38)	(0.38)
College	0.54	0.58	0.66
	(0.53)	(0.54)	(0.58)
Employee	0.03	0.50	0.17
	(0.57)	(0.45)	(0.42)
Self employed	-0.41	0.05	-0.07
	(0.86)	(0.39)	(0.41)
Civil Servant	-0.86	-0.88	-0.56
	(0.62)	(0.58)	(0.66)
POF: at most 5 times py	-0.07	-0.59	-0.18
	(0.69)	(0.59)	(0.53)
POF: at most 10 times py	-0.16	-0.58	-0.05
	(0.80)	(0.62)	(0.63)
POF: more than 10 times py	-0.35	-0.32	0.02
	(0.66)	(0.69)	(0.66)
Morning			-0.26
			(0.29)
Duration			-0.02
			(0.01)
Constant	11.92***	13.74***	13.30^{***}
	(1.81)	(2.63)	(4.72)
Adj.R2	0.00	0.22	0.29
No.Clusters	64	64	64
No.Observations	1805	1805	1805
Individual characteristics	\checkmark	\checkmark	\checkmark
Fixed effects: province, dow		\checkmark	\checkmark
Interview controls			\checkmark

Table 3 Correlation between $News_{d,p}$ and the main individual observable characteristics

Notes: robust standard errors in parentheses clustered for the day of the interview. For the definition of the control variables see Table 2; the categorical dummies omitted in the regressions are *Junior School*, *Unemployed or out of the labour force*, *POF: never*; the acronym *dow* indicates the day of the week fixed effects; the set of interview controls includes the variables *Morning* and *Duration*, a linear time trend starting from the first day of the interviews, and the enumerators fixed effects. Significance level: *** p < 0.01, ** p < 0.05, * p < 0.1 **Source**: Italian Survey of Household Income and Wealth (2014) and Corruption News Database (CND).

 Table 4

 Baseline results: the effect of corruption news exposure on corruption perceptions

		Corruption		1	Social Norn	ıs	Justi	Justice Effectiveness		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	
News	0.26**	0.28**	0.17**	0.21	0.21	0.06	-0.07	-0.06	-0.11**	
	(0.10)	(0.11)	(0.07)	(0.15)	(0.16)	(0.11)	(0.08)	(0.07)	(0.06)	
Female	× /	2.20	2.27	. ,	-0.05	0.36	()	1.40	0.97	
		(1.56)	(1.42)		(1.58)	(1.42)		(1.11)	(1.21)	
Age		-0.17**	-0.12**		-0.18***	-0.19^{***}		-0.04	-0.04	
		(0.07)	(0.06)		(0.06)	(0.06)		(0.04)	(0.04)	
High School		-5.16^{***}	-3.82^{**}		-0.61	-1.29		-1.13	0.32	
		(1.69)	(1.69)		(1.93)	(1.86)		(0.99)	(1.08)	
College		-5.67***	-3.59		-0.84	-0.11		-2.33	-0.08	
		(2.02)	(2.21)		(2.74)	(2.84)		(1.44)	(1.69)	
Employee		-4.34^{*}	-2.80		-4.39^{*}	-2.25		-0.71	-1.02	
		(2.19)	(2.18)		(2.29)	(2.12)		(1.63)	(1.56)	
Self employed		-1.55	-1.42		-3.02	-1.70		-4.28**	-1.78	
		(2.57)	(2.33)		(2.74)	(2.59)		(1.83)	(1.67)	
Civil Servant		-1.25	-4.84**		-3.01	-5.34^{*}		2.65	2.40	
		(2.20)	(2.37)		(2.82)	(2.73)		(1.87)	(1.93)	
POF: at most 5 times py		-1.08	0.33		1.58	3.62		4.32^{***}	2.77	
		(3.00)	(2.76)		(3.79)	(3.44)		(1.59)	(1.73)	
POF: at most 10 times py		2.91	3.14		5.16	4.61		4.99^{**}	1.30	
		(3.20)	(3.32)		(4.03)	(4.15)		(1.94)	(1.82)	
POF: more than 10 times py		6.84^{**}	3.37		9.20^{**}	6.54		8.62***	3.31	
		(3.03)	(3.02)		(3.99)	(4.08)		(2.05)	(2.00)	
Constant	33.17^{***}	45.28^{***}	38.70^{**}	46.43^{***}	56.13^{***}	81.80***	16.69^{***}	14.44***	-16.31^{*}	
	(1.17)	(6.31)	(16.44)	(1.87)	(6.67)	(22.79)	(1.13)	(3.47)	(8.89)	
Adj.R2	0.00	0.03	0.28	0.00	0.02	0.30	0.00	0.01	0.26	
N.Clusters	64	64	64	64	64	64	64	64	64	
N.Observations	1805	1805	1805	1805	1805	1805	1805	1805	1805	
Individual characteristics		\checkmark	\checkmark		\checkmark	\checkmark		\checkmark	\checkmark	
Fixed effects: province, dow			\checkmark			\checkmark			\checkmark	
Interview controls			\checkmark			\checkmark			\checkmark	

Notes: robust standard errors in parentheses clustered for the day of the interview. For the definition of the control variables see Table 2; the categorical dummies omitted in the regressions are *Junior School*, *Unemployed or out of the labour force*, *POF: never*; the acronym *dow* indicates day of the week fixed effects; the set of interview controls includes the variables *Morning* and *Duration*, a linear time trend starting from the first day of the interviews, and the enumerators fixed effects. Significance level: *** p<0.01, ** p<0.05, * p<0.1 **Source**: Italian Survey of Household Income and Wealth (SHIW, 2014) and Corruption News Database (CND).

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Panel A					Corrupt	ion			
Corruption news	0.18**	0.15^{**}	0.06**	2.95^{*}	0.15**	0.17**	0.19^{***}	0.22^{**}	0.16^{**}
I. I	(0.08)	(0.07)	(0.02)	(1.63)	(0.07)	(0.07)	(0.07)	(0.11)	(0.07)
Individual attitude towards tax evasion	(0.00)	(0.01)	(0.0=)	(1.00)	(0.01)	(0.01)	(0.01)	(0.22)	-3.09***
									(1.11)
Constant	38.56^{**}	55.53^{**}	55.21**	39.29**	11.14	39.03**	768.10***	534.15	49.02***
	(16.43)	(23.50)	(23.54)	(16.41)	(12.94)	(16.40)	(287.07)	(1015.70)	(16.92)
Adj.R2	0.28	0.28	0.28	0.28	0.27	0.28	0.28	0.31	0.28
Panel B					Social No	rms			
Corruption news	0.05	0.05	0.02	0.93	0.04	0.06	0.08	0.19	0.05
••••••F•••••	(0.11)	(0.11)	(0.04)	(2.25)	(0.11)	(0.11)	(0.10)	(0.15)	(0.10)
Individual attitude towards tax evasion	(-)	(-)	()	(-)	(-)	(-)	()	()	-2.43*
									(1.25)
Constant	81.98***	40.55	40.50	82.08***	49.07**	81.93***	297.65	-854.77	89.91***
	(22.75)	(24.63)	(24.68)	(22.74)	(22.71)	(22.77)	(508.24)	(1437.50)	(22.88)
Adj.R2	0.30	0.30	0.30	0.30	0.30	0.30	0.30	0.33	0.30
Panel C				In	etice Effect	tineneee			
Corruption news	-0.11*	-0.11**	-0.03*	-2 65**	_0 13**	-0.11*	-0.08	-0.10*	-0.11*
Contribution news	(0.06)	(0.05)	(0.02)	(1.14)	(0.06)	(0.06)	(0.06)	(0.06)	(0.06)
Individual attitude towards tax evasion	(0.00)	(0.00)	(0.02)	(1.14)	(0.00)	(0.00)	(0.00)	(0.00)	1.26
individual attitude towards tax evasion									(0.80)
Constant	-16 29*	33 22*	33 29*	-16.33*	-2.14	-16 56*	309.22	402.00	-20 52**
Constant	(8.95)	(18.53)	(18.57)	(8.81)	(5,55)	(8.83)	(192.05)	(465.03)	(9.20)
Adi B2	0.26	0.26	0.26	0.26	0.26	0.26	0.26	0.29	0.26
N Clusters	64	62	63	64	46	64	64	64	64
N.Observations	1805	1803	1803	1805	1653	1805	1805	1805	1805
Excluding outliers	1								
Three days moving average of News	•	1							
Cumulative version of News		•	./						
Weighted version of News			•	1					
Excluding interviews on Saturday and Sunday				•	1				
Morning interviews linked to $News_{(d-1)}$					•	\checkmark			
Adding week fixed effects						,	1		
Adding week by province fixed effects							-	1	
Control for individual attitude towards tax evasion								•	\checkmark

Table 5 Sensitivity and robustness checks

Notes: robust standard errors in parentheses clustered for the day of the interview. All the specifications include the control variables and fixed effects as in the baseline specification for Table 4, columns (3), (6) and (9); for the definition of the control variables see Table 2. The specification in column (1) excludes the two newspapers with the highest numbers of corruption news items (i.e. the outliers *Il Fatto Quotidiano* and *La Stampa*); the specification in column (2) replaces the variable $News_{dp}$ with its moving average over three days (centered on the day of the interview); the specification in column (3) replaces the variable $News_{dp}$ with its cumulative sum starting from the day of the interview and including the two days prior; the specification in column (4) replaces the variable $News_{dp}$ with a weighted version using weights equal to each newspaper's total circulation share (total circulation includes copies sold, free copies and subscriptions based on data published for the year 2013 by *Accertamenti Diffusione Stampa srl*; the specification in column (5) excludes interviews conducted on Saturday and Sunday; the specification in column (6) replaces the variable $News_{dp}$ with $News_{(d-1)p}$ for interviews started in the morning and concluded before 11 PM; the specification in column (9) adds a control variable capturing the individual's general attitude towards tax evasion. Significance level: *** p<0.01, ** p<0.05, * p<0.1 **Source**: Italian Survey of Household Income and Wealth (SHIW, 2014) and Corruption News Database (CND).

Table 6Potential mechanisms: 'facts' versus 'claims'

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	Corruption			5	locial Norms	Justice Effectiveness			
Corruption news	0.17^{*}	0.09	0.30***	0.08	0.08	-0.02	-0.09	-0.06	-0.03
	(0.09)	(0.13)	(0.08)	(0.11)	(0.16)	(0.12)	(0.06)	(0.09)	(0.08)
Constant	76.77***	57.76^{***}	23.57	80.64***	119.89^{***}	27.43	-1.85	-6.97	16.67
	(17.36)	(16.49)	(29.32)	(14.95)	(21.93)	(21.99)	(7.76)	(9.05)	(17.83)
Adj.R2	0.29	0.30	0.31	0.31	0.32	0.31	0.27	0.28	0.29
No.Clusters	62	57	57	62	57	57	62	57	57
No.Observations	1704	1437	1456	1704	1437	1456	1704	1437	1456
Baseline excluding 2 days with both facts and claims	\checkmark			\checkmark			\checkmark		
Excluding 5 days with only claims		\checkmark			\checkmark			\checkmark	
Excluding 5 days with major facts			\checkmark			\checkmark			\checkmark

Notes: robust standard errors in parentheses clustered for the day of the interview. All the specifications include the control variables and fixed effects used in the baseline specification for Table 4, columns (3), (6) and (9); for the definition of the control variables see Table 2. The specifications in columns (1), (4) and (7) repeat the baseline estimates excluding two days in which both corruption facts and claims were recorded; the specifications in columns (2), (5) and (8) exclude the five days with the highest occurrences of corruption claims; the specifications in columns (3), (6) and (9) exclude the five days with the highest occurrences of corruption facts. Significance level: *** p < 0.01, ** p < 0.05, * p < 0.1 Source: Italian Survey of Household Income and Wealth (SHIW, 2014) and Corruption News Database (CND).

	(1)	(2)	(3)	(4)	(5)	(6)
Panel A			Corru	ption		
Convictions	-0.27	-0.06	-1.49	-0.25	-0.24	-0.22
	(0.89)	(0.52)	(1.65)	(0.88)	(0.88)	(0.87)
Corruption news					0.17^{**}	0.17^{**}
					(0.07)	(0.07)
Constant	40.82^{**}	40.89**	750.78***	40.85^{**}	38.59^{**}	38.61^{**}
	(16.09)	(16.09)	(275.98)	(16.08)	(16.42)	(16.42)
Adj.R2	0.28	0.28	0.28	0.28	0.28	0.28
Panel B			Social	Norms		
Convictions	1.18	0.49	0.19	1.12	1.19	1.13
	(0.87)	(0.48)	(1.95)	(0.89)	(0.88)	(0.89)
Corruption news					0.06	0.06
					(0.11)	(0.11)
Constant	83.19^{***}	83.14***	286.07	83.10***	82.36***	82.27***
	(22.69)	(22.80)	(507.27)	(22.69)	(22.81)	(22.81)
Adj.R2	0.30	0.30	0.30	0.30	0.30	0.30
Panel C			Justice Eff	fectiveness		
Convictions	-0.96	-0.30	-0.17	-1.01	-0.98	-1.04
	(0.68)	(0.41)	(1.33)	(0.69)	(0.68)	(0.69)
Corruption news	. ,		. ,	. ,	-0.12**	-0.12**
					(0.06)	(0.06)
Constant	-18.30^{**}	-18.16^{**}	321.14^{*}	-18.28**	-16.77^{*}	-16.74*
	(8.41)	(8.44)	(192.05)	(8.41)	(8.85)	(8.84)
Adj.R2	0.26	0.26	0.26	0.26	0.26	0.26
No.Clusters	64	64	64	64	64	64
No.Observations	1805	1805	1805	1805	1805	1805
$Convictions_{dc}$	\checkmark				\checkmark	
Cumulative version of $Convictions_{dc}$		\checkmark				
Three days moving average of $Convictions_{dc}$			\checkmark			
Correction for interviews on Saturday or Sunday				\checkmark		\checkmark

Table 7Convictions as a proxy for corruption events

Notes: robust standard errors in parentheses clustered for the day of the interview. All the specifications include the control variables and fixed effects used in the baseline specification for Table 4, columns (3), (6) and (9); for the definition of the control variables see Table 2. The variable *Convictions*_{dc} indicates the number of individuals convicted because of corruption-related crimes in day d and court district c. the specifications in columns (1) and (5) use the baseline *Convictions*_{dc}; the specifications in columns (2) and (3) use variants of the *Convictions*_{dc} corresponding to the ones defined for the variable *News*_{dp} in Table 5; the specifications in columns (4) and (6) link interviews on Saturdays and Sundays to data on convictions from the preceding Friday. Significance level: *** p<0.01, ** p<0.05, * p<0.1 **Source**: Italian Survey of Household Income and Wealth (SHIW, 2014) and Corruption News Database (CND).

	(1)	(2)	(3)	(4)	(5)	(6)	
	Civil	Servant	Self-Em	ployed	Frequent Govt User		
	Yes	No	Yes	No	Yes	No	
Panel A			Corr	ruption			
Corruption news	-0.43	0.19^{**}	0.95	0.13^{*}	-0.02	0.23^{**}	
	(0.49)	(0.08)	(0.80)	(0.08)	(0.16)	(0.10)	
Constant	57.45	38.07	122.83^{**}	57.78^{**}	61.40^{**}	80.98**	
	(55.70)	(26.59)	(59.90)	(22.74)	(26.93)	(35.93)	
Adj.R2	0.27	0.29	0.21	0.28	0.30	0.29	
No.Clusters	47	61	39	64	57	60	
No.Observations	185	1620	136	1669	665	1140	
Panel B			Socia	l Norms			
Corruption news	0.35	0.77	0.77	0.05	-0.21	0.12	
	(0.69)	(1.06)	(1.06)	(0.11)	(0.19)	(0.11)	
Constant	-4.82	184.62^{**}	184.62^{**}	42.79^{*}	103.31^{***}	111.49^{***}	
	(76.86)	(70.03)	(70.03)	(24.66)	(34.27)	(33.28)	
Adj.R2	0.12	0.27	0.27	0.30	0.37	0.27	
No.Clusters	47	39	39	64	57	60	
No.Observations	185	136	136	1669	665	1140	
Panel C			Justice E	Effectivene	\$\$		
Corruption news	-0.61	-0.04	0.20	-0.12^{**}	-0.14	-0.04	
	(0.72)	(0.06)	(0.88)	(0.06)	(0.12)	(0.07)	
Constant	69.39	-24.08^{**}	7.55	31.35^{*}	55.95^{**}	-2.60	
	(48.60)	(9.75)	(44.43)	(18.57)	(22.45)	(38.24)	
Adj.R2	0.21	0.28	0.09	0.27	0.30	0.29	
No.Clusters	47	61	39	64	57	60	
No.Observations	185	1620	136	1669	665	1140	

 Table 8

 Heterogeneity across individual characteristics

Notes: robust standard errors in parentheses clustered for the day of the interview. All the specifications include the control variables and fixed effects used in the baseline specification for Table 4, columns (3), (6) and (9); for the definition of the control variables see Table 2. The *Frequent Govt User* is defined according to whether the individual declares that he/she visits general government offices more than five times per year (i.e the third and fourth category the POF variable as defined in Table 2). The control variable corresponding to the variable used for the sample split is excluded from the regression. Significance level: *** $p_i0.01$, ** $p_i0.05$, * $p_i0.1$ Source: Italian Survey of Household Income and Wealth (SHIW, 2014) and Corruption News Database (CND).

	(1)	(2)	(3)	(4)	(5)	(6)		
		Corruption						
	Civil Servant		Self-En	ployed	Frequent	PA User		
	Yes	No	Yes	No	Yes	No		
Panel A: excluding main facts								
Corruption news	0.23	0.31^{***}	0.80	0.27^{***}	0.11	0.31^{**}		
	(0.99)	(0.09)	(1.18)	(0.10)	(0.17)	(0.14)		
Constant	-3.82	9.58	9.10	28.27	36.86^{***}	41.22^{***}		
	(83.99)	(36.83)	(100.22)	(28.85)	(12.18)	(9.56)		
Adj.R2	0.22	0.33	0.26	0.32	0.35	0.32		
No.Clusters	40	54	34	57	50	53		
No.Observations	155	1301	119	1337	537	919		
Panel B: excluding main claims								
Corruption news	0.56	-0.01	1.34	0.01	-0.27	0.02		
	(1.34)	(0.12)	(1.46)	(0.12)	(0.29)	(0.18)		
Constant	115.39	-4.93	118.53	29.14	65.67^{***}	67.53^{***}		
	(87.09)	(26.66)	(144.54)	(21.56)	(15.83)	(10.91)		
Adj.R2	0.05	$0.33^{'}$	0.49	0.32	0.38	0.29		
No.Clusters	40	54	34	57	50	53		
No.Observations	155	1301	119	1337	537	919		

Table 9 Heterogeneity across individual characteristics: facts and claims

Notes: robust standard errors in parentheses clustered for the day of the interview. All the specifications include the control variables and fixed effects as in the baseline specification of Table 4, columns (3), (6) and (9); for the definition of the control variables see Table 2. The *Frequent Govt User* is defined according to whether the individual declares that he/she visits a government office more than five times per year (i.e the third and fourth category the POF variable as defined in Table 2). The main facts and main claims days that are excluded in Panels A and B, respectively, refer to the definitions provided in Table 6. The control variable corresponding to the variable used for the sample split is excluded from the regression. Significance level: *** p < 0.01, ** p < 0.05, * p < 0.1 Source: Italian Survey of Household Income and Wealth (SHIW, 2014) and Corruption News Database (CND).

A Appendix: Data

In this Appendix we provide detailed information on the characteristics of the databases used for this paper: the SHIW, the Corruption News Database (CND) and the data on convictions provided by the Ministry of Justice.

A.1 The 2014 Intermediate Wave of the Italian Survey of Household Income and Wealth (SHIW)

The SHIW is designed by the Bank of Italy and is carried out by an independent, external firm. Since 2010 it has served as the Italian database for the European Union Household Finance and Consumption Survey (HFCS) coordinated by the European Central Bank and is conducted every two years on a representative sample of households. The survey thus contains standardized sections harmonized at the European Union level from which we can obtain certain socio-economic data on households. In the gap year between the full surveys, the Bank of Italy conducts an 'intermediate wave', which includes about one fourth of the sample of the main wave and all the core questions, as well as a section on special topics that change from year to year.

The 2014 intermediate wave included a special section on corruption. Each head of household was asked three sets of questions on corruption. The first set covered perceptions of corruptions and bad social norms that are conducive to corruption. The wording was the following: 'Imagine that a citizen contacts a government office in relation to some service. Without referring to your personal experience, indicate the probability (on a scale of 0 to 100) that each event will occur. The lower the value, the lower the probability and the higher the value, the higher the probability':

- the public officer hints that he would accept a sum of money, a favor or a gift in exchange for providing the service (*Corruption*);
- the citizen has to ask for the assistance of a friend or acquaintance who works in the government office in order to expedite the receipt of the service (*Social Norms*).

In the second set of questions, respondents were asked their perceptions about the likelihood that a corrupt public officer would be discovered and subsequently sanctioned, thereby examining their perceptions of the effectiveness of measures taken to combat corruption. Again, the respondents were asked to assign a value of between 0 and 100 to indicate the probability of each event, where low (high) values correspond to low (high) probability. More specifically, we asked: 'Indicate the probability (on a scale of 0 to 100) that each event will occur. The lower the value, the lower the probability and the higher the value, the higher the probability.'

- a corrupt public officer is discovered (*Effectiveness of Investigations*);
- a corrupt public officer who has been discovered, eventually serves a prison term (*Effectiveness of Justice*).

A.2 The Corruption News Database

On each of the 82 days between 19 January and 30 March 2014 we kept track of the number of news items appearing by searching the homepages of the 30 online newspapers reported in Figure A.1 and Table A.1. These included 15 national newspapers, 12 local newspapers and the three main sports newspapers, as defined by the data published for the year 2013 by *Accertamenti Diffusione Stampa Srl*, a firm that tracks periodical circulation figures.²² Our protocol required that we start our search at 2 PM and conclude it by 4 PM. To do this we performed a Boolean search of each newspaper's homepage using a family of words in Italian that derive from the words 'corruption' (i.e. corruption, corrupted, to corrupt), 'bribe' (i.e. bribe, to bribe) and synonymous journalistic jargon like 'embezzlement' and 'misappropriation'.²³

Figure A.1 shows the total number of corruption news items recorded in the 82-day timeframe, with the newspapers *Il Fatto Quotidiano* and *La Stampa* appearing as two outliers. Table A.1 provides general descriptive statistics. Overall, we collected 945 corruption news items over the 82 days, an average of 11.5 corruption news items per day. Of the total, 67 per cent of the items appeared in national newspapers, about 32 per cent in local newspapers, and a negligible share in the sports newspapers. We applied these descriptive statistics to the total number of days tracked because we were informed that the SHIW interviews were scheduled for this time period. The main analysis was then conducted using 64 days out of the 82 on which the interviews were actually conducted, however it was necessary to track all 82 days in order to compute additional news measures, such as the three- or five-day moving averages, the leads and the lags.

A.3 Data on convictions

Data on persons convicted of corruption-related crimes are indicated by day and by court district and include the total number of persons convicted for crimes against the public administration (Articles 314–335 of the Italian Penal Code). This encompasses is a wide range of

²²http://www.adsnotizie.it.

²³The six key words in Italian were: *corruzione*, *concussione*, *tangente*, *mazzetta*, *appropriazione indebita*, *peculato*, and words derived from these (e.g. adjectives).

Figure A.1 Corruption news by newspaper



Notes: the bar chart shows the total number of corruption news items recorded in each on-line newspaper homepage over the observational period. **Source**: the authors' calculations based on data from the Corruption News Database (CND).

corruption-related offenses, such as corruption proper, falsification of public documents, bribery, and embezzlement. These same offenses correspond to the keywords used for the Boolean search performed to construct the Corruption News Database.

Data on convictions are collected by the Ministry of Justice at the court district level. In Italy, there are 29 court districts (listed in Figure A.2). This figure falls between the numbers of Italian regions (20) and provinces (110) (data do not exist for lower levels of territorial aggregation).²⁴

²⁴The database was kindly made available by the Directorate for Statistics of the Ministry of Justice and is up-to-date as of July 2014.

	Newspaper	Corruption news items per day					
Type	Name	Total No.	Mean	sd	Max	Min	Ν
Panel	A: by newspaper						
Ν	Avvenire	22	0.27	0.47	2	0	82
L	Corriere Adriatico	19	0.23	0.45	2	0	82
Ν	Corriere della Sera	24	0.29	0.53	2	0	82
\mathbf{S}	Corriere dello Sport-Stadio	6	0.07	0.26	1	0	82
L	Corriere dell'Umbria	2	0.02	0.16	1	0	82
L	Gazzetta del Sud	19	0.23	0.45	2	0	82
L	Il Centro	50	0.61	0.73	2	0	82
Ν	il Fatto Quotidiano	122	1.49	1.4	7	0	82
L	Il Gazzettino	23	0.28	0.5	2	0	82
L	Il Gazzettino (Venezia)	17	0.21	0.46	2	0	82
Ν	il Giornale	24	0.29	0.56	2	0	82
L	Il Giornale di Sicilia	25	0.3	0.51	2	0	82
Ν	Il Giorno	25	0.3	0.51	2	0	82
Ν	Il Mattino	41	0.5	0.77	4	0	82
Ν	Il Messaggero	44	0.54	0.63	2	0	82
L	il Resto del Carlino	18	0.22	0.47	2	0	82
Ν	Il Secolo XIX	29	0.35	0.57	2	0	82
Ν	Il Sole 24 ORE	22	0.27	0.55	3	0	82
Ν	Il Tempo	41	0.5	0.71	3	0	82
L	Il Tirreno	14	0.17	0.41	2	0	82
Ν	Italia Oggi	48	0.59	0.86	4	0	82
L	La Gazzetta del Mezzogiorno	38	0.46	0.69	3	0	82
\mathbf{S}	La Gazzetta dello Sport	2	0.02	0.16	1	0	82
Ν	La Nazione	19	0.23	0.45	2	0	82
Ν	la Repubblica	50	0.61	0.86	3	0	82
Ν	La Stampa	98	1.2	1.26	5	0	82
Ν	Libero	22	0.27	0.55	2	0	82
L	L'Unione Sarda	42	0.51	0.71	2	0	82
L	Messaggero Veneto	35	0.43	0.65	3	0	82
\mathbf{S}	Tuttosport	4	0.05	0.22	1	0	82
Panel	B: by newspaper type						
Ν	National	631	7.7	5.27	22	0	82
L	Local	302	3.68	2.85	12	0	82
S	Sport	12	0.15	0.45	2	0	82
Total	All Newspapers	945	11.52	7.47	30	1	82

Table A.1 Descriptive statistics for the corruption news items by newspaper

Notes: the table shows descriptive statistics for the corruption news items registered in the 82 days between 9 January and 30 March 2014. Panel A reports the figures by newspaper, Panel B by the type of newspaper: National (N), Local (L), Sports (S). The sports newspapers have a national circulation. **Source**: Corruption News Database (CND).

Table A.2 Main corruption events: description and type

Corruption news items	Event description	Event type	Date
30	Bribery case in the Italian National Space Agency and EU Commission Report on corruption (follow-up)	FC	07/02/2014
29	'Ruby ter trial': Silvio Berlusconi (Italy's former Prime Minister) accused of corruption	F	23/01/2014
29	Settlement of new Prime Minister with declarations on corruption contrast	С	18/02/2014
28	Bribery case in the Italian National Space Agency (follow-up) and EU Commission Report (follow-up)	FC	06/02/2014
27	Issuing of the EU Commission Report about corruption in the EU Member States	С	04/02/2014
23	Embezzlement cases in the Piedmont Governing Body	F	16/01/2014
21	'Ruby ter trial'(follow-up)	F	24/01/2014
21	EU Commission Report on corruption (follow-up)	С	03/02/2014
21	Settlement of new Prime Minister (follow-up)	С	19/02/2014
21	Embezzlement and bribery in the Lombardy Governing Body	F	04/03/2014
19	Embezzlement case for a member of the national Parliament and corruption case for the major of Verona	F	17/02/2014
19	Corruption case in the Lombardy public health system	F	13/03/2014
18	EU Commission Report on corruption (follow-up)	С	22/01/2014

Notes: the table shows the total number of corruption news items, the description of the corresponding event and its type (C stands for claims, F stands for facts, FC stands for cases in which news items on corruption facts overlap with those on corruption claims); *follow-up* means that corruption news item reported on that day refers to an event that happened on a previous day. **Source**: the Corruption News Database (CND).

Figure A.2 Persons convicted for corruption-related crimes by court district



Notes: the bar chart shows the total number of persons convicted for corruption-related crimes by court district over the observation period January–March 2014). **Source**: the authors' calculations based on data from the Directorate of Statistics of the Ministry of Justice.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1015 Inflation, financial conditions and non-standard monetary policy in a monetary union. A model-based evaluation, by Lorenzo Burlon, Andrea Gerali, Alessandro Notarpietro and Massimiliano Pisani (June 2015).
- N. 1016 *Short term inflation forecasting: the M.E.T.A. approach*, by Giacomo Sbrana, Andrea Silvestrini and Fabrizio Venditti (June 2015).
- N. 1017 A note on social capital, space and growth in Europe, by Luciano Lavecchia (July 2015).
- N. 1018 Statistical matching and uncertainty analysis in combining household income and expenditure data, by Pier Luigi Conti, Daniela Marella and Andrea Neri (July 2015).
- N. 1019 Why is inflation so low in the euro area?, by Antonio M. Conti, Stefano Neri and Andrea Nobili (July 2015).
- N. 1020 Forecaster heterogeneity, surprises and financial markets, by Marcello Pericoli and Giovanni Veronese (July 2015).
- N. 1021 Decomposing euro area sovereign spreads: credit,liquidity and convenience, by Marcello Pericoli and Marco Taboga (July 2015).
- N. 1022 Testing information diffusion in the decentralized unsecured market for euro funds, by Edoardo Rainone (July 2015).
- N. 1023 Understanding policy rates at the zero lower bound: insights from a Bayesian shadow rate model, by Marcello Pericoli and Marco Taboga (July 2015).
- N. 1024 Accessorizing. The effect of union contract renewals on consumption, by Effrosyni Adamopoulou and Roberta Zizza (July 2015).
- N. 1025 Tail comovement in option-implied inflation expectations as an indicator of anchoring, by Sara Cecchetti, Filippo Natoli and Laura Sigalotti (July 2015).
- N. 1026 Follow the value added: bilateral gross export accounting, by Alessandro Borin and Michele Mancini (July 2015).
- N. 1027 On the conditional distribution of euro area inflation forecast, by Fabio Busetti, Michele Caivano and Lisa Rodano (July 2015).
- N. 1028 *The impact of CCPs' margin policies on repo markets*, by Arianna Miglietta, Cristina Picillo and Mario Pietrunti (September 2015).
- N. 1029 European structural funds during the crisis: evidence from Southern Italy, by Emanuele Ciani and Guido de Blasio (September 2015).
- N. 1030 Female employment and pre-kindergarten: on the uninteded effects of an Italian reform, by Francesca Carta and Lucia Rizzica (September 2015).
- N. 1031 The predictive content of business survey indicators: evidence from SIGE, by Tatiana Cesaroni and Stefano Iezzi (September 2015).
- N. 1032 Sovereign debt exposure and the bank lending channel: impact on credit supply and the real economy, by Margherita Bottero, Simone Lenzu and Filippo Mezzanotti (September 2015).
- N. 1033 *Does trend inflation make a difference?*, by Michele Loberto and Chiara Perricone (September 2015).
- N. 1034 *Procyclicality of credit rating systems: how to manage it*, by Tatiana Cesaroni (September 2015).
- N. 1035 *The time varying effect of oil price shocks on euro-area exports*, by Marianna Riggi and Fabrizio Venditti (September 2015).
- N. 1036 Domestic and international macroeconomic effects of the Eurosystem expanded asset purchase programme, by Pietro Cova, Patrizio Pagano and Massimiliano Pisani (September 2015).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- A. MERCATANTI, A likelihood-based analysis for relaxing the exclusion restriction in randomized experiments with imperfect compliance, Australian and New Zealand Journal of Statistics, v. 55, 2, pp. 129-153, TD No. 683 (August 2008).
- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, v. 11, 2, pp. 433-465, **TD No. 709 (May 2009).**
- F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).**
- D. DOTTORI, S. I-LING and F. ESTEVAN, *Reshaping the schooling system: The role of immigration*, Journal of Economic Theory, v. 148, 5, pp. 2124-2149, **TD No. 726 (October 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian Selection*, Journal of International Economics, v. 89, 1, pp. 96-109, **TD No. 728 (October 2009).**
- L. MONTEFORTE and G. MORETTI, *Real-time forecasts of inflation: the role of financial variables*, Journal of Forecasting, v. 32, 1, pp. 51-61, **TD No. 767 (July 2010).**
- R. GIORDANO and P. TOMMASINO, *Public-sector efficiency and political culture*, FinanzArchiv, v. 69, 3, pp. 289-316, **TD No. 786 (January 2011).**
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, v. 59, pp. 212-227, TD No. 793 (February 2011).
- F. NUCCI and M. RIGGI, *Performance pay and changes in U.S. labor market dynamics*, Journal of Economic Dynamics and Control, v. 37, 12, pp. 2796-2813, **TD No. 800 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 805 (April 2011).**
- A. ACCETTURO e L. INFANTE, Skills or Culture? An analysis of the decision to work by immigrant women in Italy, IZA Journal of Migration, v. 2, 2, pp. 1-21, TD No. 815 (July 2011).
- A. DE SOCIO, *Squeezing liquidity in a "lemons market" or asking liquidity "on tap"*, Journal of Banking and Finance, v. 27, 5, pp. 1340-1358, **TD No. 819 (September 2011).**
- S. GOMES, P. JACQUINOT, M. MOHR and M. PISANI, Structural reforms and macroeconomic performance in the euro area countries: a model-based assessment, International Finance, v. 16, 1, pp. 23-44, TD No. 830 (October 2011).
- G. BARONE and G. DE BLASIO, *Electoral rules and voter turnout*, International Review of Law and Economics, v. 36, 1, pp. 25-35, **TD No. 833 (November 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, v. 11, 5, pp. 1032-1052, **TD No. 835 (November 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, in G. Gomel, D. Marconi, I. Musu, B. Quintieri (eds), The Chinese Economy: Recent Trends and Policy Issues, Springer-Verlag, Berlin, TD No. 838 (November 2011).
- A. ANZUINI, M. J. LOMBARDI and P. PAGANO, *The impact of monetary policy shocks on commodity prices*, International Journal of Central Banking, v. 9, 3, pp. 119-144, **TD No. 851 (February 2012).**
- R. GAMBACORTA and M. IANNARIO, *Measuring job satisfaction with CUB models*, Labour, v. 27, 2, pp. 198-224, **TD No. 852 (February 2012).**
- G. ASCARI and T. ROPELE, Disinflation effects in a medium-scale new keynesian model: money supply rule versus interest rate rule, European Economic Review, v. 61, pp. 77-100, TD No. 867 (April 2012).
- E. BERETTA and S. DEL PRETE, Banking consolidation and bank-firm credit relationships: the role of geographical features and relationship characteristics, Review of Economics and Institutions, v. 4, 3, pp. 1-46, TD No. 901 (February 2013).
- M. ANDINI, G. DE BLASIO, G. DURANTON and W. STRANGE, *Marshallian labor market pooling: evidence from Italy*, Regional Science and Urban Economics, v. 43, 6, pp.1008-1022, **TD No. 922 (July 2013).**
- G. SBRANA and A. SILVESTRINI, Forecasting aggregate demand: analytical comparison of top-down and bottom-up approaches in a multivariate exponential smoothing framework, International Journal of Production Economics, v. 146, 1, pp. 185-98, TD No. 929 (September 2013).

A. FILIPPIN, C. V, FIORIO and E. VIVIANO, *The effect of tax enforcement on tax morale*, European Journal of Political Economy, v. 32, pp. 320-331, **TD No. 937 (October 2013).**

2014

- G. M. TOMAT, *Revisiting poverty and welfare dominance*, Economia pubblica, v. 44, 2, 125-149, **TD No. 651** (December 2007).
- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, v.46, 4, pp. 693-713, **TD No. 730 (October 2009).**
- G. MICUCCI and P. ROSSI, *Il ruolo delle tecnologie di prestito nella ristrutturazione dei debiti delle imprese in crisi*, in A. Zazzaro (a cura di), Le banche e il credito alle imprese durante la crisi, Bologna, Il Mulino, TD No. 763 (June 2010).
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di politica economica, v. 105, 1, pp. 301-321, **TD No. 787 (January 2011).**
- R. BRONZINI and E. IACHINI, Are incentives for R&D effective? Evidence from a regression discontinuity approach, American Economic Journal : Economic Policy, v. 6, 4, pp. 100-134, TD No. 791 (February 2011).
- P. ANGELINI, S. NERI and F. PANETTA, *The interaction between capital requirements and monetary policy*, Journal of Money, Credit and Banking, v. 46, 6, pp. 1073-1112, **TD No. 801 (March 2011).**
- M. BRAGA, M. PACCAGNELLA and M. PELLIZZARI, *Evaluating students' evaluations of professors,* Economics of Education Review, v. 41, pp. 71-88, **TD No. 825 (October 2011).**
- M. FRANCESE and R. MARZIA, Is there Room for containing healthcare costs? An analysis of regional spending differentials in Italy, The European Journal of Health Economics, v. 15, 2, pp. 117-132, TD No. 828 (October 2011).
- L. GAMBACORTA and P. E. MISTRULLI, *Bank heterogeneity and interest rate setting: what lessons have we learned since Lehman Brothers?*, Journal of Money, Credit and Banking, v. 46, 4, pp. 753-778, **TD No. 829 (October 2011).**
- M. PERICOLI, *Real term structure and inflation compensation in the euro area*, International Journal of Central Banking, v. 10, 1, pp. 1-42, **TD No. 841 (January 2012).**
- E. GENNARI and G. MESSINA, How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data, International Tax and Public Finance, v. 21, 2, pp. 324-344, TD No. 844 (January 2012).
- V. DI GACINTO, M. GOMELLINI, G. MICUCCI and M. PAGNINI, *Mapping local productivity advantages in Italy: industrial districts, cities or both?*, Journal of Economic Geography, v. 14, pp. 365–394, TD No. 850 (January 2012).
- A. ACCETTURO, F. MANARESI, S. MOCETTI and E. OLIVIERI, Don't Stand so close to me: the urban impact of immigration, Regional Science and Urban Economics, v. 45, pp. 45-56, TD No. 866 (April 2012).
- M. PORQUEDDU and F. VENDITTI, Do food commodity prices have asymmetric effects on euro area inflation, Studies in Nonlinear Dynamics and Econometrics, v. 18, 4, pp. 419-443, TD No. 878 (September 2012).
- S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics, v. 76, 3, pp. 389-410, **TD No. 879 (September 2012).**
- F. D'AMURI and G. PERI, *Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession,* Journal of the European Economic Association, v. 12, 2, pp. 432-464, TD No. 886 (October 2012).
- M. TABOGA, *What is a prime bank? A euribor-OIS spread perspective*, International Finance, v. 17, 1, pp. 51-75, **TD No. 895 (January 2013).**
- G. CANNONE and D. FANTINO, *Evaluating the efficacy of european regional funds for R&D*, Rassegna italiana di valutazione, v. 58, pp. 165-196, **TD No. 902 (February 2013).**
- L. GAMBACORTA and F. M. SIGNORETTI, *Should monetary policy lean against the wind? An analysis based on a DSGE model with banking*, Journal of Economic Dynamics and Control, v. 43, pp. 146-74, **TD No. 921 (July 2013).**
- M. BARIGOZZI, CONTI A.M. and M. LUCIANI, Do euro area countries respond asymmetrically to the common monetary policy?, Oxford Bulletin of Economics and Statistics, v. 76, 5, pp. 693-714, TD No. 923 (July 2013).
- U. ALBERTAZZI and M. BOTTERO, *Foreign bank lending: evidence from the global financial crisis*, Journal of International Economics, v. 92, 1, pp. 22-35, **TD No. 926 (July 2013).**

- R. DE BONIS and A. SILVESTRINI, *The Italian financial cycle: 1861-2011*, Cliometrica, v.8, 3, pp. 301-334, **TD No. 936 (October 2013).**
- G. BARONE and S. MOCETTI, *Natural disasters, growth and institutions: a tale of two earthquakes, Journal of Urban Economics, v. 84, pp. 52-66, TD No. 949 (January 2014).*
- D. PIANESELLI and A. ZAGHINI, *The cost of firms' debt financing and the global financial crisis*, Finance Research Letters, v. 11, 2, pp. 74-83, **TD No. 950 (February 2014).**
- A. ZAGHINI, Bank bonds: size, systemic relevance and the sovereign, International Finance, v. 17, 2, pp. 161-183, TD No. 966 (July 2014).
- G. SBRANA and A. SILVESTRINI, *Random switching exponential smoothing and inventory forecasting,* International Journal of Production Economics, v. 156, 1, pp. 283-294, **TD No. 971 (October 2014).**
- M. SILVIA, Does issuing equity help R&D activity? Evidence from unlisted Italian high-tech manufacturing firms, Economics of Innovation and New Technology, v. 23, 8, pp. 825-854, TD No. 978 (October 2014).

2015

- M. BUGAMELLI, S. FABIANI and E. SETTE, *The age of the dragon: the effect of imports from China on firm-level prices*, Journal of Money, Credit and Banking, v. 47, 6, pp. 1091-1118, **TD No. 737** (January 2010).
- G. BULLIGAN, M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors,* International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- A. CIARLONE, *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- D. FANTINO, A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, v. 1, 2, pp. 219-251, TD No. 884 (October 2012).
- G. BARONE and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916 (June 2013).**
- P. ALESSANDRI and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- R. AABERGE and A. BRANDOLINI, *Multidimensional poverty and inequality*, in A. B. Atkinson and F. Bourguignon (eds.), Handbook of Income Distribution, Volume 2A, Amsterdam, Elsevier, TD No. 976 (October 2014).
- V. CUCINIELLO and F. M. SIGNORETTI, *Large banks,loan rate markup and monetary policy,* International Journal of Central Banking, v. 11, 3, pp. 141-177, **TD No. 987 (November 2014).**
- M. FRATZSCHER, D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**
- A. NOTARPIETRO and S. SIVIERO, Optimal monetary policy rules and house prices: the role of financial frictions, Journal of Money, Credit and Banking, v. 47, S1, pp. 383-410, TD No. 993 (November 2014).
- R. ANTONIETTI, R. BRONZINI and G. CAINELLI, *Inward greenfield FDI and innovation*, Economia e Politica Industriale, v. 42, 1, pp. 93-116, **TD No. 1006 (March 2015).**

FORTHCOMING

- G. DE BLASIO, D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds*, Industrial and Corporate Change, **TD No. 792 (February 2011).**
- A. DI CESARE, A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, **TD No. 831 (October 2011).**
- E. BONACCORSI DI PATTI and E. SETTE, *Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register*, Journal of Financial Intermediation, **TD No. 848** (February 2012).
- M. MARCELLINO, M. PORQUEDDU and F. VENDITTI, Short-Term GDP Forecasting with a mixed frequency dynamic factor model with stochastic volatility, Journal of Business & Economic Statistics, **TD No. 896 (January 2013).**
- M. ANDINI and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, **TD No. 915 (June 2013).**
- J. LI and G. ZINNA, *On bank credit risk: sytemic or bank-specific? Evidence from the US and UK*, Journal of Financial and Quantitative Analysis, **TD No. 951 (February 2015).**
- A. L. MANCINI, C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, **TD No. 958 (April 2014).**
- L. BURLON, *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory, **TD** No. 961 (April 2014).
- A. BRANDOLINI and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, **TD No. 965 (July 2015).**